STABILISATION PROPERTIES OF A SURE-LIKE EUROPEAN UNEMPLOYMENT INSURANCE



BANCO DE ESPAÑA Eurosistema

Documentos de Trabajo N.º 2419

Daniel Alonso

STABILISATION PROPERTIES OF A SURE-LIKE EUROPEAN UNEMPLOYMENT INSURANCE

STABILISATION PROPERTIES OF A SURE-LIKE EUROPEAN UNEMPLOYMENT INSURANCE ()

Daniel Alonso

BANCO DE ESPAÑA

(*) The views expressed in this paper are those of the author and do not necessarily represent the views of the Banco de España or the Eurosystem.

Documentos de Trabajo. N.º 2419 May 2024

https://doi.org/10.53479/36654

The Working Paper Series seeks to disseminate original research in economics and finance. All papers have been anonymously refereed. By publishing these papers, the Banco de España aims to contribute to economic analysis and, in particular, to knowledge of the Spanish economy and its international environment.

The opinions and analyses in the Working Paper Series are the responsibility of the authors and, therefore, do not necessarily coincide with those of the Banco de España or the Eurosystem.

The Banco de España disseminates its main reports and most of its publications via the Internet at the following website: http://www.bde.es.

Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

© BANCO DE ESPAÑA, Madrid, 2024

ISSN: 1579-8666 (on line)

Abstract

To moderate the falls in production and income that affect certain states or regions, countries and monetary unions have risk-sharing mechanisms. These mechanisms work by stabilising household incomes such that fluctuations in production do not filter through to consumption. Almost all existing monetary unions are true insurance unions, except for the euro area. This entails lower resilience to economic shocks and, as demonstrated during the COVID-19 crisis, implies that the ability to respond to different shocks may differ between countries and, therefore, hinder economic convergence and homogeneous operation of the euro area. In this regard, the creation of a European Unemployment Insurance (EUI) scheme is often cited as an important step towards macroeconomic smoothing within the euro area that could help mitigate the economic and social impact of large economic shocks. In this paper, I propose an EUI scheme, with partial coverage, calibrated to the characteristics of the Temporary Support to Mitigate Unemployment Risks in an Emergency (SURE scheme) introduced during the COVID-19 crisis, and test its cyclical properties through simulation exercises, based on the payment and contribution flows in each country. This paper shows that such a transfer system with a relatively limited size could make a significant contribution to stabilising economic developments, cushioning part of the disruptions in times of crisis.

Keywords: macroeconomic stabilisation, monetary union, unemployment insurance, risk-sharing.

JEL classification: F45, E63, E62, E24.

Resumen

Para moderar las caídas en la producción y en los ingresos que afectan a determinados Estados o regiones, los países y las uniones monetarias disponen de mecanismos de reparto de riesgos. Estos mecanismos funcionan estabilizando las rentas de los hogares de forma que las fluctuaciones de la producción no se filtren en el consumo. Prácticamente, todas las uniones monetarias disponen de dichos mecanismos, excepto la zona euro, lo que supone una menor capacidad de resistencia a los shocks económicos. Como quedó demostrado en la crisis del COVID-19, esto implica que la capacidad de reacción ante distintos shocks puede diferir entre países y, por tanto, dificultar la convergencia económica y un funcionamiento homogéneo de la zona. En este sentido, la creación de un sistema de seguro de desempleo europeo se cita a menudo como un paso importante hacia la estabilización macroeconómica dentro de la zona euro que podría ayudar a mitigar el impacto económico y social de las grandes crisis económicas. En este trabajo se propone un sistema de seguro de desempleo, con cobertura parcial, ajustado a las características del Apoyo Temporal para Mitigar los Riesgos de Desempleo en una Emergencia (SURE, por sus siglas en inglés), introducido durante la crisis del COVID-19, y se evalúan sus propiedades cíclicas mediante ejercicios de simulación. Este trabajo muestra cómo un sistema de transferencias de este tipo, con un tamaño relativamente limitado, podría contribuir significativamente a estabilizar la economía, amortiguando parte de las perturbaciones en tiempos de crisis.

Palabras clave: estabilización macroeconómica, unión monetaria, seguro de desempleo, reparto de riesgos.

Códigos JEL: F45, E63, E62, E24.

1 Introduction

To moderate the falls in production and income that affect certain states or regions, countries and monetary unions have risk sharing mechanisms. These mechanisms work by stabilizing household incomes in such a way that fluctuations in production do not filter into consumption. On the one hand, the agents affected by the shocks can obtain labour or financial income from other states or regions where the effect is smaller (income channel), or they can smooth their consumption or investment through access to credit (credit channel). On the other hand, among mechanisms of a public nature, fiscal transfers from a central or federal budget can also act as buffers. In the case of other federal systems, such as that of the United States, it is estimated that this type of transfers, materialized through different federal programs, have effectively mitigated the impact of idiosyncratic shocks in the different states or regions (*Burriel et al.*, 2020). However, the European Monetary Union (EMU) lacks automatic fiscal stabilizers that allow risk sharing among member states (*Banco de España*, 2017).

As has been demonstrated in the Covid-19 crisis, this implies that the ability to react to different shocks may differ between countries and, therefore, hinder economic convergence and a homogeneous operation of the area. Enhanced risk-sharing in the euro area is also backed by empirical evidence that suggests a more limited absorption of country-specific shocks in the euro area compared to the United States, among other factors, also due to the lack of a federal budget (see, e.g. *Cimadomo et al.*, 2018). Thus, a EUI has been seen by many observers and policy makers as a necessary instrument to strengthen the architecture of the euro area.

In the aftermath of the euro area crisis, the European Commission started to argue that EMU needs both a fully-fledged Banking Union and automatic fiscal stabilisers. The Five Presidents' Report published in 2015 called for the introduction of a supranational automatic stabiliser and in its Reflection Paper on the deepening of the Economic and Monetary Union of 31 May 2017, the European Commission referred to a European unemployment reinsurance scheme (EUIS) as a possible instrument that would provide a stabilisation function. In parallel, a number of Member States have expressed support for establishing an unemployment (re)insurance scheme at European level. However, implementing a European unemployment insurance may not be politically feasible and no progress has been made regarding the implementation of these proposals. European countries differ in their unemployment rates not only because they are, at a given point in time, in different phases of the business cycle, but also because they have structurally different labour market institutions. As a consequence, long-term averages of unemployment rates vary substantially across Europe. The differences in labour market institutions may imply very different optimal benefit schemes, making it potentially difficult to agree on one common system.

In this context, in April 2020 the European Commission proposed the creation of a European instrument for temporary Support to mitigate Unemployment Risks in an Emergency (SURE) to address the drastic increase in temporary unemployment related to the Covid-19 crisis and mitigate its negative economic consequences. SURE is a reinsurance system of the national unemployment insurance based on the granting of loans. In a sense, SURE can be seen as a complement to 'normal' unemployment insurance: it adds 'job insurance' in the context of a specific temporary emergency,

created by a large-scale and exogenous disaster. It was presented as a form of temporary and emergency implementation of a "European unemployment re-insurance system" and can be seen as a first step towards a permanent EUI.

This paper quantifies the benefits for the EMU derived from setting up a EUI scheme with a size (per decade) and operation similar to those of SURE. Thus, it assess the stabilising effects of a EUI in a realistic scenario where the EUI acts as an addition to national unemployment insurance schemes without replacing them and only kicks in during severe recessions and not in normal times. In particular, it uses the NiGEM macroeconometric model to perform counterfactual simulations to measure the impact of a EUI on consumption, GDP and unemployment in eleven euro area Member States covering the period 2000Q1 to 2021Q4. The analysis provides support for this European policy, which can make a significant contribution to stabilising economic developments, cushioning part of the disturbances in times of crisis. It could both lower the amplitude of national cycles and increase the synchronisation of cycles across member countries.

While the potential risk-sharing gains that can be achieved in the Eurozone by insuring severe shocks have been shown already by a number of studies (including some simulating a EUIS with NiGEM), the paper contributes to the literature by assessing whether stabilisation results from previous studies also remain robust in the wake of shocks like the Covid pandemic and by linking the SURE debate to the EUIS debate. The remainder of this paper is structured as follows. It begins with a discussion of the rationale for a supranational automatic stabilisation mechanism and the role of a European Unemployment (re)Insurance on this. Section 3 introduces the SURE instrument and the role of Short-time work schemes. Section 4 delves into the designing and modelling of a EUI, outlining general concepts and policy choices. Section 5 outlines the specific design, scenarios and trigger mechanisms examined in this paper and presents the contributions and pay-outs derived from them. Section 6 presents the results on macroeconomic stabilisation. Finally, Section 7 gives the conclusions.

2 A EUIS as an automatic stabiliser

The financial turbulence and severe economic downturn of the late 2000s and early 2010s highlighted the insufficient ability of available instruments to absorb large economic shocks in the euro area, including in Member States with low levels of public debt and seemingly sound public finances. In the euro area, Member States no longer have the exchange rate and a national monetary policy to use as tools to counter the effects of asymmetric shocks. Only fiscal policy is 'decentralised' and in principle can respond to country-specific shocks. This arrangement appears viable for normal times but it is confronted with critical problems whenever large economic disruptions arise.

Market failure is the main reason why Monetary Unions can benefit from stabilization tools (*Goodhart and Smith*, 1993; *Allard et al.*, 2013). One area where this has become especially evident is labour: with monetary policy and exchange rates absent from the EMU countries' toolboxes, labour mobility and wage flexibility should play a role, at least in theory, as important mechanisms for absorbing asymmetric shocks (*Arpaia et al.*, 2016). However, the European experience is quite different. Although recent research shows that labour mobility has increased in times of crisis, the

rate is still much lower in Europe than in the United States (*Alcidi and Thirion*, 2016). Another reason, which became especially evident during the financial crisis, is current account imbalances. *Pasimeni* (2015) argues that the EMU is characterized by an inherent deflationary bias, resulting from the absence of a supranational tool to guide demand and the fact that adjustments at the national level cannot happen through deflationary processes alone. While a banking union can help with some of these problems, an automatic stabiliser is needed. The last argument concerns spillover effects. When a country in a monetary union experiences a shock, this creates a negative spillover effect to other economies (*Majocchi and Rey*, 1993; *Frankel and Rose*, 1998; *Kalemli-Ozcan et al.*, 2003; *Allard et al.*, 2013).

These questions draw attention to the EMU architecture and the idea of establishing a common stabilization function in the EMU, recognized by the Four and Five Presidents' Reports" (2012; 2015) and more recently by the European Fiscal Board (2018). Nevertheless, the debate about such supranational automatic stabilizers for Europe is not new. It dates back to the 1970s (Marjolin, 1975; MacDougall, 1977) and it was revived in the early 1990s, when consensus was reached on the need for European financial stability (Padoa-Schioppa, 1987; Emerson, 1992). Over the past few decades, a variety of potential stabilisers have been examined. Beblav' et al. (2015) presents an overview of these mechanisms, including regional policy, public investment, debt management and taxation. For example, a regional policy proposal can be found in the report by MacDougall (1977). MacDougall (1977) developed a cyclical of cyclical grants to local or regional governments and Padoa-Schioppa (1987) proposed a similar idea. Related work by Drèze et al. (2014) focuses on public investment. Investments in social housing, renewable energy and transportation would be valuable to lagging regions. Another proposal has been made by Enderlein et al. (2013). These authors introduce the idea of a European debt agency that could issue its own bonds. Similar proposals have been discussed by De Grauwe and Moesen (2009), Allard et al. (2013) and Pisani-Ferry et al. (2013). Enderlein et al. (2013) further discussed the idea of setting up a European fund. Finally, several proposals for a common European unemployment insurance system have been put forward. One of the earliest proposals dates back to 1975, when the report by Marjolin et al. Marjolin (1975) introduced the EUIS as a tool for fiscal policy, stabilization and redistribution. This report paved the way for later work, such as that by Italianer and Vanheukelen (1993), Dullien (2013, 2014), Beblavý and Maselli (2014), Dolls et al. (2014), Jara and Sutherland (2014), Beblavý et al. (2015) and more recently by Dullien et al. (2017) and Dolls (2020).

Certainly, one of the most studied and promising mechanisms proposed so far as a supranational automatic stabiliser, both at the academic and institutional level, is a European unemployment insurance scheme (EUIS). Since stabilization is achieved automatically (i.e. net transfers from the scheme that react to national unemployment rate), the scheme has the capacity to offset fluctuations in a country's business cycle, including both expansion periods and recessions, supposedly working in a countercyclical way. This scheme may be particularly relevant to the EMU, where market failures (i.e., nominal price and wage rigidity) and low labour mobility impose an adjustment burden on workers. Since EUIS is based on unemployment, it has the advantage that unemployment benefits are counter-cyclical and highly sensitive to shocks (unemployment benefits are automatic and rapid, giving the unemployed a boost to support consumption spending). Unemployment benefits also have a high multiplier effect (*Zandi*, 2008; *Vroman*, 2010). In addition, the unemployment rate

is easily measured because it is regularly observed and monitored using survey and administrative data. Finally, it can be designed to mainly target large and possibly asymmetric shocks, which could prevent or, at least, cushion part of the shocks such as the recent Great Recession.

Strengthening national schemes by expanding their coverage and/or generosity is a particularly important channel through which stability maybe achieved. However, improving the existing national schemes would not be sufficient to achieve the same stabilisation results as a EUIS as it ignores the other two channels: NUIS cannot use a cross-country shock adjustment and may face financial or institutional constraints (EUIS can issue debt in most cases). Moreover, improving NUIS could be politically and financially difficult without a common safety net. Finally, such improvement must go hand in hand with common minimum requirements, in order to avoid Member States curtailing their budget-saving plans.

This paper contributes to the literature on measuring the stabilisation capacity of a European unemployment insurance scheme. The measurement approach, however, may affect the estimated impact of the scheme (Beblavý et al., 2015). In this regard, a first issue is whether considering the impact of the insurance scheme over the business cycle or during a recession. The stabilisation capacity of the scheme generally is larger in the latter case. A second issue is whether to use the long-term GDP trend as a benchmark to weigh the stabilisation impact of the scheme against or whether past GDP is more suitable. Estimates usually are larger when past GDP is used. Another key question is whether to focus on the average country or on the country most strongly hit. Dullien (2013) uses the long-term GDP trend and focuses on the stabilisation impact at times of recession. He obtains an effect of 11% for the average country when the net balance of payments is used as a measure (vis- a-vis the change in aggregate income in the country). Italianer and Vanheukelen (1993) used the same approach in their paper and arrive at an impact of 20% across Europe. Dolls et al. (2014) have also used a similar approach, but rely on the net sum of contributions paid and benefits received, divided by the sum of individuals' changes in employment income. Interestingly, these authors detected a stabilisation impact of 23–31% for Greece, Ireland, Italy, Spain and Portugal in the recent crises. Beblavý and Maselli (2014) and Beblavý et al. (2015) have criticised this approach, because it neglects fiscal multiplier effects. Nevertheless, the stabilisation impact reported in the literature seems to be important. Complementing this literature, this paper shows the potential risk-sharing gains that can be achieved in the Eurozone by insuring severe shocks.

3 The SURE instrument

The European Union (EU) reacted swiftly and with a wide array of instruments to the economic crisis created by the global spread of COVID-19 (*Cuadro Sáez et al.*, 2020; *Alonso Soto et al.*, 2021). SURE was a novel mechanism created to cover the sudden increase in public spending devoted to preserve employment through short-time work (STW) schemes (ERTEs in Spain) and other similar mechanisms for self-employed workers. The instrument's total envelope amounted to €100 billion. If a member state experienced a sudden and severe increase in actual and planned public expenditure for the preservation of employment because of its response to the COVID-19

pandemic, it could request financial assistance under the SURE instrument to cover part of this additional expenditure.

Although SURE was a temporary instrument (it concluded in December 2022), this initiative made operational in the short term a previous proposal from the Commission, referring to a European unemployment reinsurance system, with similar characteristics but of a permanent nature. Therefore, SURE is a reinsurance system of the national unemployment insurance based on the granting of loans. Its operation is similar to that of the federal account (Trust Fund) in the United States federal unemployment system. In this system, states accumulate funds in individual accounts through social contributions, but if they run out of funds (as happened in the vast majority of states in the financial crisis), there is a federal account that can make loans to state accounts. However, in the US, the federal account is also responsible for transferring unemployment benefits in the event of a crisis, both automatically and discretionally (*Albrizio et al.*, 2017).

While SURE was introduced in the context of unemployment reinsurance, probably the best characterisation of SURE is that it is a job insurance scheme (*Fernandes and Vandenbroucke*, 2020). It is a safety net for jobs, but not for the unemployed. The distinction is meaningful. In any existing unemployment insurance scheme, cash (and not a loan) is received by the unemployed individual. This is not the case with SURE. Thus, the missing element remains the EU capacity to top up national unemployment insurance funds in the circumstances of a recession. While the rise of joblessness can be massive, related funding will continue to rely entirely on national resources (*Andor*, 2020). Nevertheless, to the extent that SURE helps to lower the number of actual unemployed, the national unemployment benefit schemes will cope better.

SURE did not only bring a new budgetary tool to the EU but also a new way of raising and providing resources. It did not require any upfront cash contributions from member states. To back the lending scheme, member states committed irrevocable and callable guarantees worth &25 billion to the EU budget, with each guarantee calculated on the basis of their respective share of EU gross national income. Such a system ensured a high credit rating, enabling the European Commission to contract borrowings on the financial markets at favourable conditions, with the purpose of lending them to the member state requesting financial assistance. In order to avoid excessive concentration, a &60 billion limit on the total exposure to the three Member States representing the largest share of the loans was imposed.¹ The conditions of the loans, in terms of interest rate or maturity period, were set on a case-by-case basis.

The size of the instrument is key for stabilisation but there is one limit to SURE's firepower: the total amount and the aforementioned $\bigcirc 60$ billion limit in the share of loans granted to the three Member States representing the largest share. An analysis of spending on unemployment benefits provides some indication to assess this figure. Over the last decade, spending on unemployment benefits reached a record level of $\bigcirc 193$ billion in 2010 for the entire EU-27. In 2017, it stood at $\bigcirc 174$ billion. If we only consider Spain and Italy, their expenditure on unemployment benefits was equal to $\bigcirc 58$ billion in 2010 and fell to $\bigcirc 49$ billion in 2017.² These figures show that, although the

¹See Article 9 of the SURE regulation in https://eur-lex.europa.eu/legalcontent/EN/TXT/?uri=CELEX:32020R0672

²EUROSTAT, ESSPROS data

amount is significant, might not be enough to cover all unemployement benefits, particularly for some countries, but this is not the main objective of the instrument.

4 The simulated EUIS

This paper analyses stabilising and budgetary effects of an EUIS intended to provide counter-cyclical stabilization in case of large labour market shocks and designed in a way that closely mimics the SURE instrument. In particular, it asses the stabilising effects of a EUI under three (politically feasible) scenarios where the EUI acts as an addition to national unemployment insurance schemes without replacing them and only kicks in during severe recessions and not in normal times. I follow a macro-econometric approach to perform counterfactual simulations to measure the impact of such an EUIS on consumption, GDP and unemployment in eleven euro area Member States covering the period 2000Q1 to 2021Q4. In this section, the characteristics of the simulated re-insurance as well as the modelling design are presented in more detail.³

4.1 Design and characteristics

The simulated EUIS in this paper incorporates the following elements:

- Re-insurance scheme activated according to a pre-defined, unemployment and/or GDP- based trigger;
- partial coverage of negative shocks affecting national unemployment agencies, with support proportioned to the size of the shock, and a cap on cumulative balance at the country level;
- it is based on loans. The repayment of these loans is set as a percentage of the total outstanding debt each year, without interest.
- as in SURE, these loans are financed with debt issued by the EU.

These characteristics follow an insurance logic. The EUIS would cover only parts of the shocks, those causing sudden and severe rises in the unemployment rate, through the reinsurance of national unemployment benefits at a time when pressure on public finances is at the highest. Stabilizing the rest of the shocks is left at the country level. The objective is to finance occasionally high pay-outs from the scheme through loans. These loans are financed by EU debt, but it does not accumulate assets in good times: as long as the pay-offs are not activated by crisis-hit countries, the scheme is inactive and its assets do not increase.

The re-insurance scheme simulated in this paper has attractive features compared to a genuine unemployment insurance scheme. In a genuine variant of the euro area unemployment insurance scheme, the central fund pays unemployment benefits directly and continuously to all eligible euro

 $^{^{3}}$ See Annex A for a description on the main challenges related to the implementation of an EUBS, together with some solutions proposed in the literature.

area unemployed workers. Since such a scheme would be active in all periods, not restricted to large shocks, potential moral hazard would be higher. Such a scheme would also likely require further harmonisation of heterogeneous national unemployment insurance systems in terms of duration, eligibility and replacement ratios (*Dolls et al.*, 2014). In the absence of further harmonisation, a genuine system would have to be aligned to the lowest common denominator in terms of unemployment insurance, limiting its stabilisation capacity. On the contrary, an unemployment re-insurance scheme can succeed even with limited convergence, as it only re-insures national agencies according to a pre-defined transfer amount.

Figure 1 presents a basic graphical representation of the EUIS simulated in this paper. There is a borrowing capacity at the supranational level to create a European fund which makes transfers to countries through loans that have to be repaid. These pay-outs and repayment of loans have an impact on households' real disposable income. Since the model takes into account interactions between economies, other euro area countries benefit indirectly from transfers paid to another country through foreign demand.



Figure 1: EUI scheme in this paper

Source: Author's own elaboration

4.2 Scenarios

In this paper the EUIS is simulated under three different scenarios (Table 1). The simulated scenario has important consequences not only on the size and cost of the EUI but also on its stabilisation properties as well as on the long-term economic performance of participant countries. In the first contrafactual scenario, national STW schemes would have not been in place. This means that a

large share of workers in STW would have become unemployed and covered by the EUIS. As a result a contrafactual unemployment rate is generated assuming that some of dismissed workers become inactive (Figure 3).⁴ In the second contrafactual scenario, national STW schemes are in place and the EUI covers both short-term unemployed and workers in STW. In this scenario, the number of eligible unemployed is slightly larger (Figure 2) and EUI acts as the SURE instrument covering all workers in STW schemes (in addition to the short-term unemployed). Finally, in scenario 3, as in most studies in the literature, the EUI only covers short-term unemployed.⁵

In normal times and in countries with small or no STW schemes in place, both the number of eligible unemployed and the unemployment rate are similar in all three scenarios. However, at the onset of the 2008-2009 financial crisis in those countries where STW schemes already represented a standard tool to preserve employment during a temporary shortfall of demand (i.e Germany, Italy and France) and during the COVID-19 crisis in all Member States, the number of eligible unemployed and the unemployment rate differ significantly by scenario. This is particularly the case during the first quarters of the COVID-19 crisis where the number of people in short-time work schemes reached an historical record (Figure 2).

	National STW in place?	Coverage	New unemploy- ment rate?
Scenario 1	No	Short-term unemployed + new unemployed from STW	Yes
Scenario 2	Yes	Short-term unemployed + workers in STW	No
Scenario 3	Yes	only short-term unemployed	No

Table 1. Decharlos simulated in this pape	Table 1:	Scenarios	simulated	in	this	paper
---	----------	-----------	-----------	---------------	------	-------

⁴The unemployment rate in absence of STW schemes assumes that some of dismissed workers become inactive and is based on quarterly transition rates from unemployment into inactivity. For Germany, transitions rates are not available; instead, transitions rates of Denmark are used, as the two countries have similar inactivity rates.

 $^{{}^{5}}$ Scenario 2 is considered the baseline scenario and all results presented in the paper refers to this scenario (unless otherwise stated). Results for scenarios 1 and 3 are available in online annex.



Source: Author's own calculations based on Eurostat



Figure 3: Unemployment rate by scenario (%)



4.3 Triggering

One of the most critical design options of a EUIS is the trigger mechanism as it determines the frequency of the transfers and the size and power of the EUIS. It is is composed of a trigger (or indicator) variable and a threshold. The trigger variable measures the magnitude of fluctuations in employment. It enters the activation rule determining under which circumstances a pay-out from the stabilization fund is triggered. Most proposals rely on indicators such as the unemployment rate or the short-term unemployment rate. A threshold value in the activation rule stipulates how large the size of the shock must be for a pay-out to be triggered. It can either apply to the quarter-on-quarter change in the trigger variable or to the deviation from its long-term moving average.

In this paper, for the underlying indicator variable entering the activation and the contribution rule, two potential cyclical indicators are considered: the unemployment rate and the Gross Domestic Product. The reason is that both capture well the impact of the shock by focusing on the distance from a certain level. The downside of this choice is the difficulty of setting a benchmark which is, to a certain extent, discretionary. For pay-outs to be triggered, threshold values in the activation rule of 0.25 points for the quarter-on-quarter increase in the unemployment rate and 1 percentage point for the quarter-on-quarter drop in GDP are considered (Table 2).

	Trigger variable	Activation rule
Trigger 1	Unemployment rate	Double condition: quarter-on-quarter increase
		in the unumployment rate exceeds 0.25 p.p. and
		unemployment rate above its 5-year moving av-
		erage.
Trigger 2	GDP growth	Quarter-on-quarter drop in GDP exceeds 1p.p.
Trigger 3	Unemployment rate and	Trigger $1 + $ trigger 2
	GDP growth	

Table 2: Trigger mechanisms in this paper

While stronger counter-cyclical effects might be achieved by focusing on the change in the unemployment rate or GDP only, the double (triple) conditions considered in this paper, especially since they incorporate a threshold, ensure with a great degree of assurance that the Member State is indeed confronted with a large shock with a temporary and country specific element. By adding the requirement that the level of the unemployment rate must be above its five-year moving average, allow support to be less often distributed and not to provide support when the economic shock is deemed minor. This condition is needed to put countries on par, irrespective of their permanent (structural) level of unemployment. Overall, the double condition is intended to ensure that contributions (transfers) are only paid in upturns (downturns) so that pro-cyclical effects are to be avoided to the greatest possible extent. Regarding the contribution rule, it is restrictive enough to avoid pro-cyclical effects as member states are not being forced to make contribution payments in recessions that are not severe enough to trigger a transfer from the re-insurance.



Figure 4: Share of EA countries meeting the trigger condition (%)

Source: Author's own calculations based on Eurostat

The frequency of activation depends both on the scenario and, particularly, on the trigger mechanism. Figure 4 shows the share of EA countries that would have met the trigger condition for the three trigger mechanisms proposed in this paper, over the period 2000-2021 and under three different scenarios. In particular, during this period, with trigger 1 the re-insurance would have been activated most often in the period 2008–2013 and, to a lower degree, in the early 2000s, with four peaks of support (covering over 60% of the Member States) in 2009, 2012, 2020 and 2021 under scenarios 1 and 2. On the other hand, triggers 2 and 3 are more selective and leave the adjustment to larger shocks under the sole responsibility of Member States and therefore EUIS provides support less often. Under scenarios 1 and 2, EUIS would have provided support for more than 20% of EA member states only in 2008, 2009, 2011 and 2020 with trigger 2 and 3. The double trigger condition in trigger 3 makes it even more selective, triggering payouts to only 60% of EA member States at the peak of the financial crisis (2009). Nevertheless, all three triggers activate payouts to a large share of EA countries when it is really needed (2009 and 2020).

While before the COVID-19 crisis, the scenario barely affects the share of countries receiving support by trigger, this is not the case during the pandemic due to the extended use of STW. In particular, the use of STW support reached its peak during the first wave of the COVID-19 crisis in the second quarter of 2020, affecting triggers 1 and 3 as the unemployment rate is used as indicator variable. As a result, in 2020, in scenarios 1 and 2, where alternative unemployment rates are used,

EUIS covers over 90% of EA countries regardless of the trigger used, while in scenario 3 less than 40% of countries would have received support with triggers 1 and 3 as the use of STW prevented a large increase in the actual unemployment rate.

Figure 5 shows the share of quarters each country would have met the trigger conditions for payouts over the period 2000–2021. On average, the share of activations would be 13% using the unemployment rate (trigger 1), 6% using the GDP (trigger 2) and 4% if the double trigger (trigger3) is used, indicating that the overall number of activations becomes substantially smaller if the GDP or the double trigger is used in the activation rule. Portugal would have been the member state with the highest share of activations (23%) if the overall unemployment rate had been used as indicator variable, while Greece would have met the conditions for pay-outs most often if the GDP (16%) or the double trigger had been used (12.5%). Figure 5 also shows that for several member states the indicator variable selected would have led to some notable differences with regard to the number of activations. For instance, Spain would have met the condition for pay-outs 14% of the quarters if the overall unemployment rate had been used as indicator variable and only in 3.4% of them if GDP or the double trigger had been employed. At this point, it should be noted that for payouts to be triggered, it is not only necessary to meet the requirements of the trigger mechanism, but also to comply with the loan limit (20 billion euros), which limits the number (and size) of payouts that countries can received, particularly for large countries and when the shocks are relatively close together.



Figure 5: Frequency of activation by country (%)

Source: Author's own calculations based on Eurostat

4.4 Financing and size

In the proposed scheme, countries spend on unemployment benefits which is offset by extra income received by the EUI fund. The size of the pay-out (PO) is a function of the increase in the number of unemployed, so that the amount of the support is linked to the severity of the shock. When activated, the scheme pays out to the Member State an amount equivalent to its quarter-on-quarter increase in unemployment benefit expenditures, which is proxied by the nominal compensation per employee (NC) in country i and time t times the historical average replacement rate (RR) of country i times the number of new (short-term) unemployed (Nunemployed) in country i and quarter t.

$$PO_{it} = NC_{it} \times RR_i \times Nunemployed_{it} \tag{1}$$

Such pay-outs can, in exceptional circumstances, lead to high transfers. This could prove problematic for two reasons. First, some countries might run high and growing positive positions towards the fund (i.e. they receive big pay-outs for several years), which could prove politically difficult. Second, there could be some years when unemployment agencies could have difficulties absorbing the received transfers. For these reasons, as in SURE, each country will only be able to access to the EUIF to request a maximum accumulated amount, which in the reference system is 20 billion euros.⁶ This helps preventing balances to build up to excessively high levels. It also to some extent limits one-time high pay-outs, although they still occur in some cases.

$$P_{it} = PR_{it} \times OD_{it} \tag{2}$$

The simulated scheme is based on loans, which are repaid when short-term unemployment is falling and the trigger condition is not met. The repayment of these loans (P) is set as a percentage (PR) of the total outstanding debt (OD) each quarter, without interests. In the reference system the repayment term is 10 years, which is equivalent to a quarterly payment of 2.5% of the outstanding debt. As in SURE, these loans are financed by a supranational entity (EU) that issues debt, allowing the individual risk of the Member States to be shared among them. The EU or euro area institutions have a much greater borrowing capacity than most individual governments, as it has been shown in the success of the European Commission emissions. of 'SURE social bonds', as well as of 'eurobonds' to finance NGEU programmes. Furthermore, from a modelling perspective, the capacity of EU institutions to borrow in international financial markets dilutes the distinction of the EU as an open or closed economy. While there could not be permanent current account imbalances between the EU and the rest of the world, the capacity to smooth the consequences of severe shocks would remain.

Contributions to and transfers from the European unemployment insurance fund vary significantly over time (Tables 3 and 4) and across member states (Tables 5 and 6) depending on the scenario and trigger mechanism but also on the national level of social security benefits and, particularly, on the labour market situation and economic developments. Average payouts are much higher in

⁶Although SURE regulation allows a country to request a larger amount, this condition ensures that the \bigcirc 60 billion limit of loans granted to the three Member States representing the largest share is met. Results for an alternative model with a limit based on GDP (2.5% of quarterly GDP) are shown in the Annex.

scenarios 1 and 2 as workers in STW are covered by the EUIS, while in scenario 3, only shortterm unemployed are covered by the EUIS, resulting in lower average transfers from the EUIS. In particular, averaged over the whole period, the fund transfers an average pay-out of between 0.14%and 0.27% of GDP per quarter in scenarios 1 and 2, depending on the scenario and trigger, while payouts range between 0.03% and 0.09% in scenario 3. However, a more telling figure for a scheme based on an insurance logic is the non-zero average pay out since a simple average of the pay-outs includes all the cases where there is not pay-out at all. When it does activate, the fund provides on average a pay-out between 2% and 4.7% of GDP in scenarios 1 and 2 and between 0.8% and 1.3% in scenario 3.

Focusing on the distribution of payouts over time at the EA level, table 3 shows that there are clearly two peaks of support, in 2009 and mainly in 2020 in scenarios 1 and 2. In particular, the EUIS would have transferred payouts by as much as 0.62% and 3.84% of euro area GDP in 2009 and 2020 respectively. However, in scenario 3, the EUIS would have offered support mainly at the peak of the great financial crisis (ranging from 0.44% to 0.58% depending on the trigger) and somewhat less at the peak of the pandemic (from 0.11% to 0.29%), since workers in STW would not be covered. As a comparison figure, on average from 2000 to 2013, the euro area spent 1.3% of GDP in social public expenditure related to unemployment. Conversely, repayments of payouts (contributions) would have peaked in 2021 in scenarios 1 and 2 and in all scenarios the contributions would have been relatively high and constant from 2010 onwards (Table 4).

	Av	verage Pay-c	out	Non-zero average Pay-out						
	Trigger 1	Trigger 2	Trigger 3	Trigger 1	Trigger 2	Trigger 3				
2000	0.00	0.01	0.00	0.11	0.49	-				
2001	0.06	0.01	0.01	0.59	0.37	0.33				
2002	0.08	0.00	0.00	0.63	0.11	0.03				
2003	0.12	0.03	0.02	0.74	1.14	1.58				
2004	0.12	0.01	0.00	0.68	0.17	-				
2005	0.03	0.00	0.00	0.44	-	-				
2006	0.03	0.02	0.02	0.74	1.37	1.37				
2007	0.01	0.00	0.00	0.33	0.02	-				
2008	0.23	0.19	0.11	1.47	0.95	1.34				
2009	0.53	0.62	0.47	1.15	1.69	1.97				
2010	0.09	0.02	0.02	0.57	0.55	0.55				
2011	0.15	0.06	0.06	0.75	0.78	0.91				
2012	0.09	0.01	0.01	0.41	0.17	0.17				
2013	0.12	0.03	0.03	0.72	0.95	0.95				
2014	0.03	0.00	0.00	0.57	-	-				
2015	0.04	0.03	0.00	0.60	1.24	-				
2016	0.01	0.00	0.00	0.23	0.17	-				
2017	0.00	0.00	0.00	-	0.01	-				
2018	0.00	0.00	0.00	-	-	-				
2019	0.00	0.00	0.00	-	-	-				
2020	3.84	3.28	3.30	7.89	9.60	10.91				
2021	0.46	0.03	0.03	2.50	2.51	2.51				

Table 3: Average payouts by year, EA (% GDP)

Source: Author's own calculations based on Eurostat

=

	Aver	age contrib	ution	Non-zero	average con	tribution
	Trigger 1	Trigger 2	Trigger 3	Trigger 1	Trigger 2	Trigger 3
2000	0.00	0.00	0.00	0.00	0.01	0.00
2001	0.00	0.00	0.00	0.02	0.01	0.01
2002	0.00	0.00	0.00	0.02	0.01	0.01
2003	0.01	0.00	0.00	0.04	0.02	0.01
2004	0.01	0.00	0.00	0.04	0.02	0.01
2005	0.02	0.00	0.00	0.04	0.02	0.02
2006	0.02	0.00	0.00	0.04	0.02	0.03
2007	0.02	0.00	0.00	0.04	0.02	0.02
2008	0.01	0.00	0.00	0.04	0.03	0.03
2009	0.03	0.02	0.02	0.11	0.09	0.08
2010	0.07	0.05	0.04	0.11	0.08	0.09
2011	0.05	0.04	0.03	0.10	0.08	0.08
2012	0.05	0.04	0.03	0.11	0.09	0.09
2013	0.07	0.04	0.03	0.13	0.08	0.08
2014	0.07	0.04	0.03	0.13	0.08	0.07
2015	0.06	0.04	0.03	0.11	0.07	0.07
2016	0.06	0.03	0.02	0.10	0.06	0.06
2017	0.05	0.03	0.02	0.09	0.05	0.05
2018	0.04	0.03	0.02	0.08	0.05	0.05
2019	0.03	0.02	0.01	0.08	0.04	0.05
2020	0.14	0.12	0.11	0.42	0.35	0.34
2021	0.29	0.21	0.21	0.41	0.30	0.29

Table 4: Average contributions by year, EA (% GDP)

Source: Author's own calculations based on Eurostat

A comparison of average and non-zero average pay-outs can also be made at the country level. In scenarios 1 and 2, small countries with high unemployment rates and largely impacted by both the global financial crisis and the COVID-19 crisis like Portugal, Greece or Cyprus benefit considerably from high transfers as a percentage of their GDP (Table 5). Similarly, small countries with relatively high gross wages and large STW schemes like Austria also receive relatively large average transfers as a result of very high payouts at the peak of the COVID-19 crisis. On the other hand, large countries like France, Spain and Italy benefit less due to the 20 billion limit, despite the impact of both crises on their economies. Moreover, in the case of Italy, the number of short-term unemployed relative to overall unemployment is comparatively low, which results in lower average payouts despite the very high overall level of unemployment.

For example, in scenarios 1 and 2 the average pay-out that Portugal receives from 2000 to 2021 amounts up to 0.52% of its GDP per quarter, but taking only the quarters for which it did receive a payment, the average pay out is up to 8.9%. This again is due to the insurance principle: the fund does not activate often but when it does it can provide significant payments for some countries.

For Austria, the fund would have provided average and non-average pay-outs of up to 0.30% and 11% respectively due to very large payouts at the peak of the pandemic. On the contrary, Spain would have received support for smaller amounts, so that the average and non-average pay-outs are not that different (up to 0.15% and up to 3.68% respectively). Finally, in scenario 3, cross country differences are smaller. Using the unemployment rate as indicator variable, payouts range from 0.01% (0.4% non-zero average) of GDP in Germany to 0.13% in Estonia (1.4% non-zero average), while using the GDP as indicator variable transfers vary from 0.02% in Italy (0.62% non-zero average) to 0.12% in Spain (2.7% non-zero average). Finally, if the double trigger mechanism is used, the payouts would be even lower and the mechanism would be barely activated for many member states. However, the EUIS would provide significant payments for some countries. For example, Spain would have received non-zero average payouts of 3.3% of its GDP.

	Av	verage Pay-c	out	Non-zero average Pay-out						
	Trigger 1	Trigger 2	Trigger 3	Trigger 1	Trigger 2	Trigger 3				
AT	0.30	0.26	0.26	1.99	5.62	11.00				
BE	0.26	0.22	0.22	2.25	6.33	6.33				
$\mathbf{C}\mathbf{Y}$	0.62	0.43	0.42	3.64	9.56	12.28				
DE	0.03	0.05	0.03	0.61	1.41	2.51				
EE	0.32	0.28	0.24	2.79	3.05	4.31				
\mathbf{ES}	0.14	0.12	0.12	1.02	3.45	3.45				
\mathbf{FI}	0.27	0.17	0.15	2.40	3.85	6.53				
\mathbf{FR}	0.07	0.06	0.06	0.53	1.81	1.81				
GR	0.30	0.22	0.21	1.85	1.38	1.67				
IE	0.30	0.17	0.16	2.06	2.08	3.48				
IT	0.09	0.07	0.07	0.71	2.15	3.03				
LT	0.25	0.11	0.11	1.80	1.58	1.58				
LU	0.41	0.19	0.15	3.04	2.84	4.40				
LV	0.21	0.15	0.13	2.28	1.61	2.28				
\mathbf{MT}	0.47	0.46	0.43	4.63	4.04	9.53				
NL	0.17	0.13	0.12	1.16	2.93	10.65				
\mathbf{PT}	0.52	0.32	0.30	2.27	7.00	8.90				
\mathbf{SI}	0.33	0.22	0.20	1.83	6.33	8.98				
SK	0.17	0.14	0.14	3.14	4.17					

Table 5: Average payouts by country (% GDP)

Source: Author's own calculations based on Eurostat

Turning to the aggregate cumulative balance of the system and the flows of payouts and contributions generated by the scheme at the aggregate level, figure 6 shows that the total debt accumulated at the end of the period does not vary substantially regardless of the trigger in scenarios 1 and 2, largely due to the cap on country cumulative balances. However, the choice of the trigger mecha-

	Aver	age contrib	ution	Non-zero	average cor	tribution	
	Trigger 1	Trigger 2	Trigger 3	Trigger 1	Trigger 2	Trigger 3	
AT	0.05	0.03	0.03	0.13	0.11	0.48	
BE	0.05	0.02	0.02	0.11	0.08	0.08	
CY	0.08	0.04	0.03	0.21	0.14	0.15	
DE	0.00	0.02	0.00	0.01	0.05	0.05	
$\mathbf{E}\mathbf{E}$	0.05	0.05	0.04	0.17	0.18	0.14	
\mathbf{ES}	0.06	0.05	0.05	0.13	0.13	0.13	
FI	0.05	0.03	0.02	0.19	0.14	0.10	
\mathbf{FR}	0.03	0.02	0.02	0.07	0.07	0.07	
GR	0.07	0.06	0.05	0.17	0.15	0.17	
IE	0.07	0.03	0.03	0.14	0.10	0.09	
IT	0.03	0.02	0.02	0.11	0.07	0.06	
LT	0.05	0.03	0.03	0.10	0.07	0.06	
LU	0.06	0.03	0.02	0.15	0.07	0.04	
LV	0.04	0.04	0.03	0.14	0.09	0.11	
\mathbf{MT}	0.06	0.06	0.05	0.14	0.14	0.12	
NL	0.04	0.02	0.01	0.10	0.07	0.23	
\mathbf{PT}	0.08	0.04	0.04	0.20	0.13	0.13	
\mathbf{SI}	0.05	0.02	0.02	0.13	0.08	0.40	
SK	0.02	0.01	0.01	0.04	0.04	0.27	

Table 6: Average contributions by country (% GDP)

nism has a significant impact on the flows that are generated and, therefore, on the system's ability to deal with truly large asymmetric shocks. A system based on the unemployment rate is not very selective since it does not provide much support at the peak of the financial crisis and it is activated for some countries between crisis, not allowing a quick repayment of the debt. On the other hand, if the GDP or the double trigger are used instead, the system is activated to provide support when it is really needed, allowing the amortization of the debt in periods of expansion, limiting the accumulated debt in the system and allowing a better support at the onset of the pandemic. Scenario 3 depicts a different picture in terms of the cumulative balance. When trigger 1 is used, the cumulative debt reaches over 90 billion at the peak of the global financial crisis, while it remains bellow 40 billion when the double trigger is used.

Source: Author's own calculations based on Eurostat



Source: Author's own calculations based on Eurostat

4.5 Modelling: The NiGEM model

In order to assess the macroeconomic stabilisation provided by the EUI scheme, in this paper I follow a macro-econometric approach to simulate the adoption of such a scheme. In particular, I use the NiGEM macroeconometric model developed by the National Institute of Economic and Social Research (NIESR) to perform counterfactual simulations and measure the impact of a EUI on consumption, GDP and unemployment in eleven euro area Member States covering the period 2000Q1 to 2021Q4.⁷ In a first step, payouts and contributions flows are genereated in Excel for each country. Second, these flows are feeded into NiGEM:

- Payouts are modelled as an (exogenous) shock to social transfers.
- An additional (exogenous) negative shock to government interests the size of payouts is introduced.
- Contributions are modelled as an (exogenous) shock to government interest payments.

 $^{^7{\}rm The}$ sample represents 98% of euro area GDP: Austria, Belgium, Finland, France, Germany, Greece, Ireland, Italy, Netherlands, Portugal and Spain.

- In scenario 1, an additional (exogenous) negative shock to the number of employed persons is introduced.
- Monetary policy is kept endogenous so it can react. However, results with exogenous monetary policy are also provided.

Finally, to estimate the stabilising effects, the variance of (counterfactual) consumption and GDP, trend-adjusted using a Hodrick-Prescott filter, are computed.

NiGEM is a comprehensive simulation and forecasting model for the global economy incorporating typical New-Keynesian elements such as rational expectation formation by economic agents as well as price and wage rigidities. The model enables a broad but nonetheless detailed modeling of the global economy. NiGEM models all OECD countries as well as numerous emerging nations with up to 130 equations and the aim of simulating their reaction to exogenous developments; the simulations also factor in international feedback effects—through foreign trade, for example—as well as economic policy responses—such as monetary and fiscal policy—for economic developments.

The analysis uses counterfactual simulations. The model simulates an economic development in the past where the deviation from the actual historical course results from the development of exogenous or endogenous variables which diverge from reality. The simulations are created on the basis of quarterly data and simulate the introduction of a EUI scheme for the euro area.

NiGEM captures all social welfare benefits in a single variable called "transfers". Transfers consist primarily of pensions and to a lesser extent of other social benefits, including for unemployment. Higher transfers increase households' disposable income, which is used for saving and consumption purposes. Thus, an increase in transfers mainly affects the economy through higher consumption expenditures by private households. This raises aggregate demand and – if the economy has unused resources – raises output and employment.

The extent to which higher transfer benefits increases expenditure is determined by the marginal propensity to consume. One would usually assume that most of the additional unemployment benefits will be consumed by the unemployed, as these are in any case lower than the previous earned income. In contrast one might assume a lower share of additional pension benefits being consumed by pensioners (saving for bequest motives and/or because the pension is not offsetting a temporary decline in earned income). As NiGEM does not, however, distinguish between unemployment benefits and other types of benefits, all benefits are treated equally and the marginal propensity to consume in the model is the same for unemployment benefits as it is for pensions. In fact it is the average across all households for any given increase in disposable income. This should ceteris paribus leads to underestimate the additional household consumption expenditure forthcoming from an increase in unemployment benefits in NiGEM.

5 Modelling results

The analysis in this section is limited to the eleven EA member states included in the NiGEM model. Simulations conducted using the National Institute Global Economic Model (NiGEM) show that had a common unemployment insurance scheme existed in the euro area since the creation of the Monetary Union in 1999, the cyclical fluctuations in some countries would have been considerably less pronounced. During periods of economic downturn, a European unemployment insurance would bolster disposable income and consequently stabilize consumption which, in turn, would have a stabilizing effect on production and employment in the countries affected. The loss of purchasing power in countries with strong economic growth would also have a stabilizing impact by cooling down the overheating economy. The impact would vary from country to country depending on the generosity of the insurance system; the trigger mechanism and the scenario.

5.1 Economic impact

Looking at the entire euro area, a European unemployment insurance contribute to an increase in macroeconomic stability. In times of economic downturn, as in 2009 or 2020, a European unemployment insurance system would have raised households' disposable income, thus reducing consumption losses and ultimately mitigating the negative effects on growth. Moreover, income stabilisation in one country also benefits other countries via external demand. As shown in Figure 7, the introduction of a European unemployment insurance scheme would have significantly reduced GDP and consumption losses in the euro area in 2009 and in 2020. In particular, in 2009 the counterfactual gain in consumption in the eleven euro area countries studied stood at 0.29%, 0.5% and 0.3% with triggers 1, 2 and 3 respectively, compared with a situation without a European unemployment insurance. In 2020, the increase in consumption would be very similar with triggers 1 and 2, at 0.41% and 0.4% respectively, while with trigger 3 it would stood at 0.5% since it allows for a better replenishment of reserves.

In terms of GDP, there is a maximum stabilising effect of approximately 0.2% of the euro area's GDP in 2009 with trigger 2 and in 2020 with trigger 3 as the proximity of the two crises prevents the replenishment of reserves with triggers 1 and 2. Overall, in the run-up to the crisis (2005 to 2007) and also during the period of economic recovery (between 2011 and 2019), contractive effects on GDP are visible. This emphasizes the counter-cyclical nature of this European unemployment insurance model (Figure 7). However, the introduction of a European unemployment insurance would lead to a slight increase in euro area's GDP at the end of the period (2021), ranging from 0.07% with model 1 and trigger 1 to 0.29% with model 3 and trigger 3 (Table 7).

Regarding its impact on employment, a EUIS would have significantly lowered the increase in unemployment at the peak of both crises (2009-2010 and 2020-2021) and at the end of the period the counterfactual unemployment rate would be lower compared to the baseline scenario without EUIS (see Figure 7 and Table 7).

At the country level, increasing transfers from a European unemployment insurance would also result in counterfactual gains in consumption, GDP and employment, particularly for member states with weak economies, which could lead to significant deviations from the baseline, i.e., economic developments without a European unemployment insurance. As shown in Figure 8 and Table 7, the introduction of a European unemployment insurance scheme would have significantly increased average consumption and GDP in several member states. This is primarily due to a less severe slump



Source: Author's own calculations based on NiGEM model

in disposable income which, in turn, moderates the downturn in consumer demand. In addition, as a result of the European unemployment insurance easing the burden on the national budget, there would be stronger growth in public spending which, in turn, would have a stabilising impact. For example, on average during the whole period, a EUIS would have increased consumption in approximately a 0.3% in Portugal and Greece in scenario 2 with trigger 1 (a slightly less average increase is obtained with triggers 2 and 3). In terms of GDP, the increase would have been more moderate, in both countries GDP would have expanded a 0.1% on average (Figure 8).

However, average results mask significant differences in the time profile of the effects of a EUIS by country. While during the COVID-19 crisis average increase in GDP and consumption would be higher in small countries like Belgium, Portugal and Greece, large countries like France, Italy and Spain would have benefited more during the global financial crisis (Figure 9). This is the result of the design of the mechanism. In model 1, the 20 billion limit of the system prevents large countries to receive more support during the pandemic, lowering the average increases in consumption and GDP and at the same time allows small countries to receive very large transfers at the peak of the COVID-19 crisis, which lead to large gains in both consumption and GDP in those countries. In model 2, which doubles the size and the limit, the counterfactual gains are larger for all countries but small countries still would have larger gains during the pandemic. However, when the system

Table 7: Counterfactual gain/losses in consumption, GDP and unemployment (%)

Trigger 1

			Mod	del 1					Mod	lel 2			Model 3					
		2022			2039			2022			2039			2022 2039				
	С	GDP	U	C	GDP	U	С	GDP	U									
AT	-0.00	0.02	0.01	0.02	-0.01	0.00	-0.00	0.04	0.02	0.04	-0.01	0.00	0.02	0.10	0.05	0.04	-0.04	0.01
BE	1.10	0.19	-0.05	-0.46	-0.08	-0.00	2.48	0.47	-0.14	-0.90	-0.19	-0.00	0.58	0.08	0.02	-0.25	-0.03	0.01
DE	0.16	0.02	0.00	-0.00	0.01	-0.00	0.33	0.04	0.01	-0.01	0.03	-0.00	0.73	0.11	0.03	-0.06	0.06	-0.01
$\mathbf{E}\mathbf{A}$	0.15	0.02	-0.01	-0.05	0.01	-0.00	0.26	0.03	-0.01	-0.09	0.02	-0.01	0.28	0.02	0.02	-0.06	0.05	-0.01
\mathbf{ES}	0.06	0.06	-0.04	-0.08	-0.00	0.00	0.13	0.08	-0.06	-0.15	-0.00	0.00	0.23	0.12	-0.06	-0.04	-0.00	0.01
\mathbf{FI}	-0.07	-0.01	0.01	0.02	0.02	0.00	-0.11	-0.01	0.02	0.05	0.04	0.00	-0.12	-0.03	0.03	0.07	0.05	0.00
\mathbf{FR}	-0.10	-0.08	0.02	-0.03	0.02	-0.01	-0.19	-0.16	0.04	-0.05	0.06	-0.03	-0.15	-0.20	0.08	-0.08	0.10	-0.04
GR	0.56	0.14	0.20	-0.29	-0.03	-0.01	0.54	0.14	0.21	-0.28	-0.02	-0.01	0.10	-0.02	0.09	-0.11	0.02	0.00
IE	-0.04	0.02	0.00	-0.02	0.01	-0.00	-0.06	0.04	0.00	-0.04	0.01	-0.00	-0.06	0.05	0.00	-0.04	0.02	-0.00
IT	0.10	0.01	-0.01	-0.04	0.01	-0.00	0.24	0.03	-0.01	-0.09	0.02	-0.00	0.31	0.07	-0.02	-0.07	0.05	-0.00
\mathbf{NL}	-0.09	0.01	0.04	-0.00	0.01	0.00	-0.16	0.01	0.07	-0.00	0.02	0.00	-0.18	-0.03	0.12	0.01	0.02	0.01
\mathbf{PT}	2.29	0.81	-0.55	-0.37	0.01	-0.02	2.27	0.79	-0.54	-0.35	0.02	-0.02	0.33	0.07	-0.09	-0.07	0.03	0.00

Trigger 2

	Model 1								Mod	iel 2			Model 3					
		2022			2039			2022			2039			2022			2039	
	С	GDP	U	С	GDP	U	C	GDP	U	C	GDP	U	C	GDP	U	C	GDP	U
AT	-0.00	0.03	0.01	0.01	-0.00	0.00	-0.00	0.08	0.02	0.02	-0.00	0.00	0.01	0.13	0.04	0.04	-0.04	0.01
BE	1.25	0.25	-0.08	-0.44	-0.10	-0.00	2.59	0.54	-0.17	-0.88	-0.20	-0.00	0.71	0.16	0.00	-0.25	-0.03	0.01
DE	0.06	-0.03	0.01	-0.01	-0.02	0.00	0.21	0.01	0.01	-0.02	-0.02	-0.00	0.51	0.04	0.03	-0.06	0.06	-0.01
$\mathbf{E}\mathbf{A}$	0.17	0.03	-0.02	-0.07	-0.02	-0.00	0.48	0.14	-0.03	-0.13	-0.02	-0.01	0.48	0.13	-0.01	-0.06	0.05	-0.01
\mathbf{ES}	0.23	0.10	-0.04	-0.11	-0.02	-0.00	1.08	0.41	-0.13	-0.26	-0.06	-0.01	0.64	0.27	-0.08	-0.04	-0.00	0.01
\mathbf{FI}	-0.06	0.00	0.01	0.01	0.01	-0.00	-0.09	0.01	0.02	0.02	0.03	-0.00	-0.09	-0.01	0.03	0.07	0.05	0.00
\mathbf{FR}	-0.01	-0.04	-0.00	-0.08	-0.02	-0.01	0.10	-0.01	-0.01	-0.16	-0.02	-0.02	0.33	0.04	-0.00	-0.08	0.10	-0.04
GR	0.36	0.08	0.12	-0.22	-0.07	-0.01	0.35	0.11	0.13	-0.21	-0.06	-0.01	0.10	0.02	0.08	-0.11	0.02	0.00
IE	-0.04	0.03	0.00	-0.02	0.01	-0.00	-0.05	0.07	-0.00	-0.03	0.01	-0.00	-0.06	0.07	0.00	-0.04	0.02	-0.00
IT	0.30	0.12	-0.02	-0.06	-0.01	-0.00	0.86	0.38	-0.06	-0.12	-0.01	-0.01	0.93	0.42	-0.07	-0.07	0.05	-0.00
NL	-0.08	0.03	0.03	-0.01	0.01	-0.00	-0.10	0.07	0.07	-0.00	0.02	-0.00	-0.12	0.02	0.13	0.01	0.02	0.01
\mathbf{PT}	1.46	0.49	-0.32	-0.25	-0.01	-0.02	1.46	0.53	-0.34	-0.24	0.00	-0.02	0.48	0.17	-0.11	-0.07	0.03	0.00

Trigger 3

Model 1							Model 2						Model 3					
		2022			2039		2022 2039							2022		2039		
	С	GDP	U	C	GDP	U	С	GDP	U	С	GDP	U	C	GDP	U	C	GDP	U
AT	-0.00	0.04	0.01	0.02	-0.01	0.00	-0.00	0.09	0.02	0.03	-0.01	0.00	0.01	0.14	0.05	0.04	-0.04	0.01
BE	1.26	0.24	-0.07	-0.44	-0.09	-0.00	2.59	0.53	-0.16	-0.88	-0.19	-0.00	0.72	0.15	0.01	-0.25	-0.03	0.01
DE	0.19	0.03	0.00	-0.01	-0.00	0.00	0.34	0.07	0.01	-0.02	0.00	-0.00	0.63	0.11	0.03	-0.06	0.06	-0.01
$\mathbf{E}\mathbf{A}$	0.22	0.06	-0.01	-0.06	-0.00	-0.00	0.53	0.16	-0.03	-0.13	-0.01	-0.01	0.53	0.15	-0.00	-0.06	0.05	-0.01
\mathbf{ES}	0.24	0.11	-0.04	-0.10	-0.01	-0.00	1.09	0.42	-0.13	-0.25	-0.05	-0.01	0.65	0.28	-0.08	-0.04	-0.00	0.01
\mathbf{FI}	-0.05	-0.00	0.01	0.01	0.01	-0.00	-0.09	0.00	0.02	0.02	0.03	-0.00	-0.09	-0.01	0.03	0.07	0.05	0.00
\mathbf{FR}	-0.02	-0.05	0.01	-0.06	0.00	-0.01	0.10	-0.02	0.00	-0.14	-0.00	-0.02	0.33	0.03	0.00	-0.08	0.10	-0.04
\mathbf{GR}	0.37	0.09	0.12	-0.21	-0.07	-0.00	0.37	0.11	0.14	-0.20	-0.07	-0.01	0.14	0.03	0.10	-0.11	0.02	0.00
IE	-0.03	0.03	-0.00	-0.02	0.00	-0.00	-0.04	0.07	-0.00	-0.02	0.01	-0.00	-0.05	0.07	0.00	-0.04	0.02	-0.00
\mathbf{IT}	0.36	0.16	-0.03	-0.05	0.00	-0.00	0.92	0.42	-0.07	-0.12	0.00	-0.01	0.99	0.46	-0.07	-0.07	0.05	-0.00
NL	-0.06	0.02	0.04	-0.00	0.01	-0.00	-0.08	0.06	0.08	0.00	0.02	-0.00	-0.10	0.02	0.14	0.01	0.02	0.01
\mathbf{PT}	1.38	0.47	-0.31	-0.24	-0.00	-0.02	1.38	0.51	-0.32	-0.23	0.01	-0.02	0.49	0.17	-0.11	-0.07	0.03	0.00

Source: Author's own calculations based on NiGEM model

sets a limit based on the size of the economy (as in model 3) the results differ considerably and large weak economies like France, Italy and Spain but also the euro area as a whole benefit more from the introduction of a EUIS.



Source: Author's own calculations based on NiGEM model

On the other hand, the introduction of a European unemployment insurance would have had a primarily negative impact on growth in strong economies like Austria, Germany, Finland or the Netherlands for the period before the pandemic (Figure: 10 (panel a)) and particularly during the global financial crisis (Figure 9 (panel b)). The predominantly negative impact on the GDP of these countries is due to the low transfers they received from EUIS until the COVID-19 crisis as labour markets in these countries were hardly affected during the financial and subsequent debt crisis. Since other countries experienced significant increases in unemployment and received large sums of transfers from EUIS, lifting consumption and demand, monetary policy reacts raising interest rates, affecting aggregate demand and slowing down economic growth in all countries but affecting more negatively the aforementioned strong economies (Figure 10). However, during the COVID-19 crisis, all countries would have experienced stronger growth with a EUIS than without it (Figure 9 (panel b)), offsetting in most countries the negative impact prior to the pandemic (Figure 10 (panel b)).

A EUIS, therefore, allows for income and consumption smoothing over time and implies a significant reduction in consumption, GDP and employment losses during a recession. At the same time, in the years immediately preceding the COVID-19 crisis, the growth in the GDP of several member states and of the euro area as a whole resulting from the European unemployment insurance would have been slightly more moderate. The transfers from the European unemployment insurance



Figure 9: Average counterfactual gain/losses by period (%)

(a) Consumption

(b) GDP



Source: Author's own calculations based on NiGEM model



Figure 10: Average counterfactual gain/losses in GDP, active vs. non-active Monetary Policy (%)

(a) 2000-2019

(b) 2000-2021





scheme would have decreased in the pre-crisis period because unemployment was also declining. This would have subdued the expansion of private consumer demand and thus also mitigated the overheating of the economy somewhat. However, during periods of high growth—or periods of low unemployment—a European unemployment insurance would have resulted in dampening effects on the economies of the member states. Due to the growth in employment during economic upswings, contributions also increase, leading to an outflow of cash which, in turn, slows overall economic growth. Nonetheless, while a EUIS carries a cost in terms of GDP in expansionary periods, this tends to fade away and even carries counterfactual output gains, albeit minimal, in the long run (Table 7).

5.2 Stabilisation

The smoothing out of consumption, GDP and unemployment can be measured by the decline in their variance. In this section, I compare the variance of consumption, GDP and unemployment trend-adjusted using a Hodrick-Prescott (HP) filter, with and without EUIS, first within countries and then across countries.

5.2.1 Within countries

Most countries have experienced a significant increase in the volatility of their business cycles, particularly since the onset of the global financial crisis (Figure 11). Introducing a EUIS would have led to a decline in consumption, GDP and unemployment volatility for almost all countries. At the aggregate level, the variance of consumption in the euro area decreases between 5% and 9% depending on the trigger. In general, all countries to some extend benefit from introducing a EUIS, with some countries like Belgium and Greece benefiting from a decline in the variance of consumption and GDP by more than 15% and 5% respectively (Figure 12) with triggers 2 and 3 providing slightly more stabilisation overall.



Figure 11: Dispersion across time (2000-2021)

Source: Author's own calculations based on Eurostat



Figure 12: Counterfactual gain/losses in dispersion across time (%)

Source: Author's own calculations based on NiGEM model

Table 8: Counterfactual	gain	/losses	in	dispersion	across	time	(%))
-------------------------	------	---------	---------------	------------	--------	------	-----	---

Trigger 1

	Model 1								Mod	lel 2			Model 3					
	2000-2021 2000-2039						2000-2021 2000-2039						2000-2021 2000-2039				9	
	С	GDP	U	C	GDP	U	C	GDP	U	C	GDP	U	C	GDP	U	С	GDP	U
AT	0.03	-0.61	-0.27	0.03	-0.63	0.06	0.04	-1.31	-1.15	0.04	-1.35	-0.46	0.09	-2.69	-1.74	0.09	-2.78	-0.41
BE	-17.53	-6.12	-4.50	-17.60	-6.14	-5.15	-40.04	-14.60	-9.47	-40.02	-14.59	-10.74	-8.60	-7.83	-4.54	-8.64	-7.87	-5.19
DE	-4.61	-1.17	1.64	-4.83	-1.19	1.56	-11.67	-3.17	3.11	-11.90	-3.21	2.97	-40.84	-8.87	-0.61	-40.01	-8.95	-0.22
$\mathbf{E}\mathbf{A}$	-5.46	-2.65	-2.43	-5.50	-2.68	-2.65	-10.39	-5.03	-4.05	-10.43	-5.08	-4.32	-17.56	-7.09	-2.82	-17.63	-7.16	-2.86
\mathbf{ES}	-3.24	-2.38	-3.95	-3.22	-2.48	-3.86	-6.11	-4.52	-5.61	-6.07	-4.69	-5.44	-4.85	-3.96	-4.97	-4.82	-4.15	-4.82
\mathbf{FI}	-0.07	-0.29	0.56	-0.07	-0.29	0.58	-0.12	-0.79	0.42	-0.13	-0.79	0.45	-0.06	-1.38	-0.31	-0.07	-1.38	-0.29
\mathbf{FR}	-1.59	-1.54	-2.40	-1.61	-1.54	-2.43	-2.27	-3.39	-6.07	-2.32	-3.39	-6.10	-4.49	-4.66	-6.87	-4.58	-4.69	-7.00
GR	-16.39	-13.06	-0.29	-16.08	-13.08	-0.76	-16.34	-13.14	-0.31	-16.03	-13.17	-0.76	-8.32	-5.55	0.54	-8.24	-5.56	0.25
IE	-0.10	-0.02	-0.30	-0.10	-0.02	-0.23	-0.18	-0.06	-0.55	-0.18	-0.07	-0.41	-0.17	-0.13	-0.82	-0.17	-0.14	-0.56
\mathbf{IT}	-0.86	-1.30	-0.54	-0.90	-1.32	-0.62	-3.61	-2.47	-3.09	-3.67	-2.50	-3.16	-3.84	-2.87	-2.90	-3.91	-2.92	-2.99
\mathbf{NL}	-0.50	-1.49	-0.30	-0.46	-1.53	-0.31	-0.91	-3.38	-1.28	-0.84	-3.43	-1.25	-1.11	-5.87	-2.01	-0.99	-5.96	-1.53
\mathbf{PT}	-0.24	-5.88	-16.27	0.08	-6.13	-17.58	-0.29	-6.74	-16.76	0.03	-7.00	-18.01	-3.42	-4.95	-6.90	-3.45	-5.04	-7.20

Trigger 2

Model 1					Model 2					Model 3								
	2000-2021 2000-2039			9	2000-2021			2000-2039		2000-2021			2000-2039					
	С	GDP	U	C	GDP	U	C	GDP	U	С	GDP	U	C	GDP	U	С	GDP	U
AT	0.08	-0.89	-0.91	0.08	-0.90	-0.62	0.16	-1.93	-1.30	0.17	-1.96	-0.66	0.20	-3.39	-1.56	0.09	-2.78	-0.41
BE	-20.18	-7.80	-4.69	-20.24	-7.80	-5.40	-41.65	-16.86	-8.71	-41.62	-16.83	-10.15	-10.11	-10.07	-2.69	-8.64	-7.87	-5.19
DE	-7.07	-4.25	0.53	-6.82	-4.25	0.50	-21.29	-7.50	-1.20	-20.45	-7.52	-1.03	-44.16	-12.76	1.79	-40.01	-8.95	-0.22
$\mathbf{E}\mathbf{A}$	-7.27	-3.80	-3.20	-7.26	-3.82	-3.41	-15.57	-7.05	-5.26	-15.54	-7.12	-5.64	-22.83	-9.32	-3.36	-17.63	-7.16	-2.86
\mathbf{ES}	-3.44	-2.76	-2.01	-3.42	-2.88	-1.96	-7.53	-6.43	-3.54	-7.50	-6.74	-3.42	-5.69	-5.37	-3.00	-4.82	-4.15	-4.82
FI	-0.25	-0.69	0.61	-0.26	-0.69	0.61	-0.19	-1.28	0.29	-0.20	-1.28	0.29	-0.19	-1.82	-0.19	-0.07	-1.38	-0.29
\mathbf{FR}	-3.54	-2.41	-2.42	-3.58	-2.42	-2.52	-7.88	-5.07	-2.71	-7.99	-5.09	-3.00	-12.94	-7.52	0.20	-4.58	-4.69	-7.00
\mathbf{GR}	-18.55	-10.65	-0.32	-18.32	-10.65	-0.79	-18.53	-10.94	-0.33	-18.30	-10.93	-0.81	-9.26	-5.81	0.34	-8.24	-5.56	0.25
IE	-0.16	-0.03	-0.31	-0.16	-0.04	-0.25	-0.18	-0.12	-0.65	-0.18	-0.13	-0.50	-0.18	-0.18	-0.90	-0.17	-0.14	-0.56
IT	-3.47	-2.34	-3.44	-3.53	-2.37	-3.55	-8.51	-4.41	-4.22	-8.63	-4.48	-4.39	-9.04	-5.07	-4.28	-3.91	-2.92	-2.99
NL	-0.64	-2.61	-0.61	-0.61	-2.62	-0.70	-1.06	-5.40	-1.72	-1.01	-5.45	-1.81	-1.33	-7.78	-1.84	-0.99	-5.96	-1.53
\mathbf{PT}	-2.09	-6.25	-12.99	-1.99	-6.49	-13.68	-2.13	-7.67	-13.84	-2.03	-7.91	-14.40	-3.50	-6.55	-7.22	-3.45	-5.04	-7.20

Trigger 3

Model 1					Model 2					Model 3								
	2000-2021			2000-2039		:	2000-2021		2000-2039		2000-2021		2000-2039					
	С	GDP	U	C	GDP	U	C	GDP	U	C	GDP	U	С	GDP	U	C	GDP	U
AT	0.07	-1.11	-0.71	0.08	-1.13	-0.31	0.16	-2.14	-1.12	0.17	-2.18	-0.35	0.20	-3.59	-1.36	0.09	-2.78	-0.41
BE	-20.15	-8.10	-5.03	-20.21	-8.10	-5.76	-41.62	-17.11	-8.86	-41.58	-17.08	-10.30	-10.08	-10.36	-2.98	-8.64	-7.87	-5.19
DE	-15.37	-3.44	-2.55	-14.70	-3.46	-2.34	-28.52	-6.61	-3.37	-27.34	-6.64	-3.02	-49.46	-11.70	1.83	-40.01	-8.95	-0.22
$\mathbf{E}\mathbf{A}$	-9.18	-3.96	-3.25	-9.16	-3.99	-3.43	-17.38	-7.19	-5.27	-17.35	-7.27	-5.62	-24.54	-9.45	-3.33	-17.63	-7.16	-2.86
\mathbf{ES}	-3.43	-2.79	-1.98	-3.41	-2.91	-1.93	-7.52	-6.45	-3.51	-7.49	-6.77	-3.39	-5.68	-5.39	-2.97	-4.82	-4.15	-4.82
\mathbf{FI}	-0.08	-0.68	-0.18	-0.09	-0.68	-0.18	-0.02	-1.27	-0.51	-0.03	-1.27	-0.51	-0.03	-1.80	-0.99	-0.07	-1.38	-0.29
\mathbf{FR}	-3.47	-2.64	-2.77	-3.52	-2.65	-2.86	-7.81	-5.31	-3.09	-7.93	-5.33	-3.36	-12.87	-7.75	-0.14	-4.58	-4.69	-7.00
GR	-19.06	-10.92	-0.53	-18.83	-10.91	-0.99	-19.04	-11.20	-0.54	-18.81	-11.19	-1.01	-10.83	-6.63	0.05	-8.24	-5.56	0.25
IE	-0.10	-0.04	-0.37	-0.10	-0.05	-0.27	-0.12	-0.12	-0.71	-0.12	-0.14	-0.53	-0.12	-0.19	-0.96	-0.17	-0.14	-0.56
IT	-3.88	-2.30	-2.36	-3.94	-2.34	-2.52	-8.87	-4.35	-3.11	-9.00	-4.44	-3.30	-9.40	-5.02	-3.16	-3.91	-2.92	-2.99
NL	-0.60	-3.01	-1.25	-0.57	-3.04	-1.28	-1.02	-5.77	-2.35	-0.97	-5.83	-2.34	-1.29	-8.12	-2.45	-0.99	-5.96	-1.53
\mathbf{PT}	-0.15	-5.24	-12.23	-0.08	-5.47	-12.96	-0.20	-6.67	-13.10	-0.12	-6.90	-13.71	-1.47	-5.77	-7.10	-3.45	-5.04	-7.20

Source: Author's own calculations based on NiGEM model

5.2.2 Across countries

The dispersion between business cycles among euro area countries have also experienced a significant increase during downturns (Figure 13). The introduction of a EUIS would have also lowered the dispersion of real cycles in euro area countries, particularly around the COVID-19 crisis but also at the onset of the Global Financial Crisis (Figure: 14), implying that business cycles are better synchronized, even in the long term when the limit of loans is based on GDP (Model 3) instead of based on a fixed amount (Models 1 and 2) (Table 9).



Figure 13: Dispersion across countries (2000-2021)

Source: Author's own calculations based on Eurostat



Source: Author's own calculations based on NiGEM model

		2009			2020			2030	
	С	GDP	U	C	GDP	U	C	GDP	U
Model 1, trigger 1	-3.59	3.77	7.97	22.12	-10.32	-16.51	0.33	7.54	15.73
Model 1, trigger 2	23.85	-1.41	-2.77	6.31	-2.18	-3.89	-0.12	4.82	9.91
Model 1, trigger 3	-1.97	2.13	4.46	5.82	-2.01	-3.51	0.30	4.34	8.91
Model 2, trigger 1	5.59	4.17	8.73	26.34	-11.41	-17.98	1.08	6.81	14.20
Model 2, trigger 2	31.27	-1.15	-2.26	5.18	-4.70	-8.51	0.17	4.26	8.79
Model 2, trigger 3	6.20	2.39	5.04	4.96	-4.50	-8.08	0.58	3.76	7.74
Model 3, trigger 1	4.51	3.97	8.37	2.36	-3.35	-6.25	1.96	-1.60	-3.15
Model 3, trigger 2	31.25	-1.18	-2.32	-3.89	-3.46	-6.52	1.07	-0.86	-1.68
Model 3, trigger 3	6.18	2.36	4.98	-2.83	-3.24	-6.03	1.48	-1.03	-2.03

Table 9: Counterfactual gain/losses in dispersion across countries, EA11 (%)

Source: Author's own calculations based on NiGEM

6 Conclusions

This paper evaluates an inter-temporal reinsurance mechanism that closely resembles the Temporary Support to Mitigate Unemployment Risks is an Emergency (SURE), which the European Commission introduced in 2020 in response to the coronavirus outbreak, as a tool to alleviate budgetary pressure through the recessionary increase in unemployment expenditures. In particular, it analyses the counterfactual welfare gains and macroeconomic stabilisation effects of introducing such a mechanism. The analysis finds significant benefits from such an instrument, contrary to the conventional classical view, which suggests that such automatic stabilisers should have little impact on welfare due to the low costs of business cycles (*Lucas Jr*, 2003). The analysis provides support for this European policy, which can make a significant contribution to stabilising economic developments, cushioning part of the disturbances in times of crisis. It could both lower the amplitude of national cycles and increase the synchronisation of cycles across member countries. The increased stabilisation of the cycles within the euro area comes at the cost of a very slight slowdown in countries' growth trajectory at the peak of the cycle, arising from the contributions collected during higher growth periods and for strong economies also during downturns due to monetary policy action.

Finally, it should be noted that the results presented are somewhat conservative as NiGEM assumes that the unemployed have the same propensity to consume as other households. Therefore, the welfare gains presented in this paper can be seen as a lower-bound, but not out of sync with the gains that most euro area countries can achieve with a risk-sharing EUIS. On the other hand, the mechanism proposed is based on intergovernmental loans. Thus, governments might not directly transfer these funds to households (as this paper assumes) and as a result the scheme could have less impact (*Burriel et al.*, 2020). In this sense, a genuine fiscal risk-sharing mechanism at the European level, of sufficient size, would have a much greater capacity to achieve the optimal aggregate tone of fiscal policy, thus complementing the work of monetary policy. Along with these considerations, the introduction of a fiscal mechanism at the supranational level could change the incentives of national governments, so it should be integrated into a deeper fiscal union, which mitigates moral hazard among its participants.

References

- Ábrahám, Árpád, João Brogueira de Sousa, Ramon Marimon and Lukas Mayr. (2023). "On the design of a European Unemployment Insurance System". *European Economic Review*, 156. https://doi.org/10.1016/j.euroecorev.2023.104469
- Albrizio, Silvia, Juan Carlos Berganza and Iván Kataryniuk. (2017). "Federal unemployment insurance in the United States". *Economic Bulletin Banco de España*, 2/2017. https://repositorio. bde.es/handle/123456789/8896
- Alcidi, Cinzia, and Gilles Thirion. (2016). "Assessing the Euro Area's Shock-Absorption Capacity: Risk Sharing, Consumption Smoothing and Fiscal Policy". CEPS Special Report, 146, Centre for European Policy Studies, European Commission. https://papers.ssrn.com/abstract=2859428
- Allard, Céline, Petya Koeva Brooks, John Bluedorn, Fabian Bornhorst, Franziska Ohnsorge and Katharine Christopherson Puh. (2013). "Toward a Fiscal Union for the Euro Area". IMF Staff Discussion Notes, 13/09, International Monetary Fund. https://www.imf.org/external/pubs/ft/ sdn/2013/sdn1309.pdf
- Alonso, Daniel, Alejandro Buesa, Carlos Moreno Pérez, Susana Párraga Rodríguez and Francesca Viani. (2021). "Fiscal policy measures adopted since the second wave of the health crisis: the euro area, the United States and the United Kingdom". Documentos Ocasionales, 2118, Banco de España. https://repositorio.bde.es/handle/123456789/17540
- Andor, László. (2020). "Sure EU Capacity for Stabilising Employment and Incomes in the Pandemic". *Intereconomics*, 55(3), pp. 139-142. https://doi.org/10.1007/s10272-020-0888-y
- Arnold, Nathaniel, Bergljot Barkbu, Elif Ture, Hou Wang and Jiaxiong Yao. (2018). "A Central Fiscal Stabilization Capacity for the Euro Area". IMF Staff Discussion Notes, 18/03, International Monetary Fund. https://www.imf.org/-/media/Files/Publications/SDN/2018/SDN1803.ashx
- Arpaia, Alfonso, Aron Kiss, Balazs Palvolgyi and Alessandro Turrini. (2016). "Labour mobility and labour market adjustment in the EU". *IZA Journal of Migration*, 5(21). https://doi.org/10.1186/ s40176-016-0069-8
- Banco de España. (2017). "Chap. 4. Fiscal policy in the euro area". In *Annual Report* 2016. https://www. bde.es/f/webbde/SES/Secciones/Publicaciones/PublicacionesAnuales/InformesAnuales/16/ Files/cap4e.pdf
- Beblavý, Miroslav, Daniel Gros and Ilaria Maselli. (2015). "Reinsurance of National Unemployment Benefit Schemes". CEPS Working Document, 401, Centre for European Policy Studies, European Commission. https://cdn.ceps.eu/wp-content/uploads/2015/01/wd401.pdf
- Beblavý, Miroslav, Karolien Lenaerts and Ilaria Maselli. (2017). "Design of a European Unemployment Benefit Scheme". CEPS Research Report, 2017/04, Centre for European Policy Studies, European Commission. https://www.ceps.eu/ceps-publications/design-europeanunemployment-benefit-scheme/
- Beblavý, Miroslav, and Ilaria Maselli. (2014). "An Unemployment Insurance Scheme for the Euro Area: A Simulation Exercise of Two Options". CEPS Special Report, 98, Centre for European Policy Studies, European Commission. https://www.ceps.eu/ceps-publications/unemployment-insurance-scheme-euro-area-simulation-exercise-two-options/

- Bénassy-Quéré, Agnès, Markus Brunnermeier, Henrik Enderlein, Emmanuel Farhi, Clemens Fuest, Pierre-Olivier Gourinchas, Philippe Martin, Jean Pisani-Ferry, Hélène Rey, Isabel Schnabel, Nicolas Veron, Beatrice Weder Di Mauro and Jeromin Zettelmeyer (2018). "Reconciling risk sharing with market discipline: A constructive approach to euro area reform". CEPR Policy Insight, 91, Centre for Economic Policy Research. https://cepr.org/system/files/publicationfiles/103106-policy_insight_91_reconciling_risk_sharing_with_market_discipline_a_ constructive_approach_to_euro_area_reform.pdf
- Biggs, Michael, and Thomas Mayer. (2010). "The output gap conundrum". In "The Impact of the Financial Crisis on the Real Economy". *Intereconomics: Review of European Economic Policy*, 45(1), pp. 11-16. https://www.intereconomics.eu/contents/year/2010/number/1/article/theimpact-of-the-financial-crisis-on-the-real-economy.html
- Burriel, Pablo, Panagiotis Chronis, Maximilian Freier, Sebastian Hauptmeier, Lukas Reiss, Dan Stegarescu and Stefan Van Parys. (2020). "A Fiscal Capacity for the Euro Area: Lessons from Existing Fiscal-Federal Systems". ECB Occasional Paper Series, 239, European Central Bank. https://doi.org/10.2139/ssrn.3571387
- Carnot, Nicolas, Magdalena Kizior and Gilles Mourre. (2017). "Fiscal stabilisation in the Euro-Area: A simulation exercise". CEB Working Paper, 17/025, Centre Emile Bernheim, Université Libre de Bruxelles. https://dipot.ulb.ac.be/dspace/bitstream/2013/259556/3/wp17025.pdf
- Cimadomo, Jacopo, Sebastian Hauptmeier, Alessandra Anna Palazzo and Alexander Popov. (2018). "3. Risk sharing in the euro area". *Economic Bulletin European Central Bank*, 3/2018. https://www.ecb.europa.eu/pub/pdf/other/ecb.ebart201803_03.en.pdf
- Claveres, Guillaume, and Jan Stráský. (2018). "Euro Area unemployment insurance at the time of zero nominal interest rates". OECD Economics Department Working Papers, 1498, Organisation for Economic Co-operation and Development. https://doi.org/10.1787/99b92f5b-en
- Cuadro Sáez, Lucía, Fernando López Vicente, Susana Párraga Rodríguez and Francesca Viani. (2020). "Fiscal policy measures in response to the health crisis in the main euro area economies, the United States and the United Kingdom". Documentos Ocasionales, 2019, Banco de España. https://www.bde.es/f/webbde/SES/Secciones/Publicaciones/PublicacionesSeriadas/ DocumentosOcasionales/20/Files/do2019e.pdf
- Darvas, Zsolt. (2015). "Mind the gap (and its revision)!". Bruegel Blog, 20 May. https://www. bruegel.org/blog-post/mind-gap-and-its-revision
- De Grauwe, Paul, and Wim Moesen. (2009). "Gains for All: A Proposal for a Common Euro Bond". In "Common Euro Bonds: Necessary, Wise or to be Avoided?". *Intereconomics*, 44(3), pp. 132-135. https://doi.org/10.1007/s10272-009-0287-x
- Delpla, J. (2012). "A euro-wide conditional unemployment insurance". In seminar "EU Level Economic Stabilisers", Brussels, July.
- Dolls, Mathias. (2020). "An unemployment re-insurance scheme for the eurozone? Stabilizing and redistributive effects". CESifo Working Papers, 8219, Center for Economic Studies, University of Munich. https://doi.org/10.2139/ssrn.3576297
- Dolls, Mathias, Clemens Fuest, Dirk Neumann and Andreas Peichl. (2014). "An unemployment insurance scheme for the euro area: Evidence at the micro level". *Economic shock absorbers for the Eurozone Deepening the debate on automatic stabilizers*, Brussels, 20 June. https://pdfs. semanticscholar.org/ba8e/120b21c153ad156b773e6651ad670a326016.pdf

- Dolls, Mathias, Clemens Fuest, Dirk Neumann and Andreas Peichl. (2018). "An unemployment insurance scheme for the euro area? A comparison of different alternatives using microdata". *International Tax and Public Finance*, 25, pp. 273-309. http://dx.doi.org/10.1007/s10797-017-9440-9
- Drèze, Jacques H., Alain Durré and Jean-François Carpantier. (2014). "Fiscal Integration and Growth Stimulation in Europe". *Recherches Économiques de Louvain/Louvain Economic Review*, 80(2), pp. 5-45. https://doi.org/10.3917/rel.802.0005
- Dullien, Sebastian. (2007). "Improving Economic Stability in Europe: What the Euro Area Can Learn from the United States' Unemployment Insurance". Working Paper, 1/2007, Stiftung Wissenschaft & Politik. https://www.swp-berlin.org/publications/products/arbeitspapiere/ Paper_US_KS_neu_formatiert.pdf
- Dullien, Sebastian. (2012). "A European unemployment insurance as a stabilization device Selected issues". Paper prepared for brainstorming workshop on July 2, 2012 at the DG EMPL, European Commission. https://ec.europa.eu/social/BlobServlet?docId=10437&langId=en
- Dullien, Sebastian. (2013). "A euro-area wide unemployment insurance as an automatic stabilizer: Who benefits and who pays?". Paper prepared for the European Commission (DG EMPL). https://ec.europa.eu/social/BlobServlet?docId=10438&langId=en
- Dullien, Sébastien. (2014). "Preventing permanent transfers under a European Unemployment Insurance: Can a claw-back mechanism be the answer?". Presentation at the conference *Economic shock absorbers for the Eurozone: Deepening the debate on automatic stabilisers*, Brussels, 20 June.
- Dullien, Sebastian, Jonas Fernández, Marc López, Gero Maass, Daniel del Prado and Jakob von Weizsäcker. (2017). "Fit for purpose: A German-Spanish proposal for a robust European Unemployment Insurance". Friedrich-Ebert-Stiftung. https://library.fes.de/pdf-files/bueros/ madrid/17661.pdf
- Emerson, Michael. (1992). One market, one money: An evaluation of the potential benefits and costs of forming an economic and monetary union. Oxford University Press.
- Enderlein, Henrik, Jann Spiess, Lucas Guttenberg and António Vitorino. (2013). *Blueprint for a cyclical shock insurance in the euro area*. Notre Europe, Jacques Delors Institute. https://institutdelors.eu/wp-content/uploads/2020/08/blueprintforacyclicalshockinsurancene-jdisept2013-1.pdf
- Enders, Zeno, and David Vespermann. (2021). "Cross-country unemployment insurance, transfers, and trade-offs in international risk sharing". CESifo Working Paper, 8965. Center for Economic Studies, University of Munich. https://doi.org/10.2139/ssrn.3815453
- European Fiscal Board. (2018). Annual Report 2018. https://commission.europa.eu/document/ download/294f5bc7-1f30-4872-b1ac-acc7872f41ce_en?filename=2018-efb-annual-report_en.pdf
- Fernandes, Sofia, and Frank Vandenbroucke. (2020). "SURE: A welcome lynchpin for a European unemployment re-insurance". Policy Paper, 251, Notre Europe, Jacques Delors Institute. https:// institutdelors.eu/wp-content/uploads/2020/08/PP251_SURE_Fernandes-Vandenbrouck_200417_ EN-1.pdf
- Fichtner, Ferdinand, and Peter Haan. (2014). "European Unemployment Insurance: Economic Stability without Major Redistribution of Household Incomes". DIW Economic Bulletin, 17 November, pp. 39-51. https://www.diw.de/documents/publikationen/73/diw_01.c.489540.de/ diw_econ_bull_2014-10-6.pdf

- Frankel, Jeffrey, and Andrew Rose. (1998). "The Endogeneity of Optimum Currency Area Criteria". *The Economic Journal*, 108(449), pp. 1009-1025. https://doi.org/10.1111/1468-0297.00327
- Goodhart, Charles, and Stephen Smith. (1993). "Stabilization". European Economy: Reports and Studies, 5, pp. 417-455.
- Gossé, Jean-Baptiste, Camille Jehle, Yann Perdereau and Roger Vicquéry. (2022). "European unemployment insurance and macroeconomic stabilisation: Are permanent fiscal transfers between States needed?". *Bulletin de la Banque de France*, 240(3). https://publications.banque-france.fr/en/european-unemployment-insurance-and-macroeconomic-stabilisation-are-permanent-fiscal-transfers
- Grauwe, Paul De. (2003). "The Euro at Stake? The Monetary Union in an Enlarged Europe". *CESifo Economic Studies*, 49(1), pp. 103-121. https://doi.org/10.1093/cesifo/49.1.103
- Ignaszak, Marek, Philip Jung and Keith Kuester. (2020). "Federal unemployment reinsurance and local labor-market policies". IZA Discussion Papers, 13886, IZA - Institute of Labor Economics. https://docs.iza.org/dp13886.pdf
- Ince, Onur, and David H. Papell. (2013). "The (un)reliability of real-time output gap estimates with revised data". *Economic Modelling*, 33, pp.713-721. https://doi.org/10.1016/j.econmod.2013.05.023
- Italianer, Alexander, and Marc Vanheukelen. (1993). "Proposals for Community stabilization mechanisms: Some historical applications". *European Economy: Reports and Studies*, 5, pp. 493-510.
- Jara, H. Xavier, and Holly Sutherland. (2014). "The implications of an EMU unemployment insurance scheme for supporting incomes". Research Note, 3/2013, European Comission. https://ec.europa.eu/social/BlobServlet?docId=11582&langId=en
- Juncker, Jean Claude, Donald Tusk, Jeroen Dijsselbloem, Mario Draghi and Martin Schulz. (2015). "Completing Europe's economic and monetary union". European Commission. https://www. ecb.europa.eu/pub/pdf/other/5presidentsreport.en.pdf
- Kalemli-Ozcan, Sebnem, Bent E. Sorensen and Oved Yosha. (2003). "5. Economic integration, industrial specialization, and the asymmetry of macroeconomic fluctuations". In *Economic Policy in the International Economy: Essays in Honor of Assaf Razin*. Cambridge University Press, pp. 121-156. https://doi.org/10.1017/cbo9780511610141.007
- Kaufmann, Christoph, Maria Grazia Attinasi and Sebastian Hauptmeier. (2023). "Macroeconomic stabilisation properties of a euro area unemployment insurance scheme". Journal of Economic Dynamics and Control, 153(104698). https://doi.org/10.1016/j.jedc.2023.104698
- Lucas Jr., Robert E. (2003). "Macroeconomic priorities". *American Economic Review*, 93(1), pp. 1-14. https://doi.org/10.1257/000282803321455133
- MacDougall, Donald. (1977). "Report of the study group on the role of public finance in European integration". Economic and Financial Series, 13, Commission of the European Communities. https://op.europa.eu/en/publication-detail/-/publication/907775d1-93ce-4824-b0b0-cedc2389731b
- Majocchi, Alberto, and Mario Rey. (1993). "A special financial support scheme in economic and monetary union: Need and nature". *European Economy: Reports and Studies*, 5, pp. 457-480.

- Marjolin, Robert. (1975). "Report of the Study Group 'Economic and Monetary Union 1980' and Annex I". Commission of the European Communities. http://aei.pitt.edu/1009/
- Moyen, Stéphane, Nikolai Stähler and Fabian Winkler. (2019). "Optimal unemployment insurance and international risk sharing". *European Economic Review*, 115, pp. 144-171. https://doi.org/10.1016/j.euroecorev.2019.03.004
- Padoa-Schioppa, Tommaso. (1987). Efficiency, stability, and equity: Strategy for the evolution of the economic system of the European Community. Oxford University Press.
- Pasimeni, Paolo. (2015). "The Economic Rationale of an EMU Fiscal Capacity". *Proceedings of OeNB Workshop Toward a Genuine Economic and Monetary Union. Oesterreichische Nationalbank, Vienna,* 2015. https://papers.ssrn.com/sol3/papers.cfm?abstract_id=3216390
- Pisani-Ferry, Jean, Erkki Vihriälä and Guntram B. Wolff. (2013). "Options for a euro-area fiscal capacity". Bruegel Policy Contribution, 2013/01. https://www.bruegel.org/sites/default/files/wp-content/uploads/imported/publications/pc_2013_02.pdf
- Strauss, Robert, Olivier Bontout, Guy Lejeune, Magdalena Ciesielska and Roberta Di Girolamo. (2013). "Paper on automatic stabilisers". European Commission, 4 October. https://ec.europa. eu/social/BlobServlet?docId=10964&langId=en
- Thirion, Gilles. (2017). "European Fiscal Union: Economic rationale and design challenges". CEPS Working Document, 2017/01, Centre for European Policy Studies, European Commission. https://cdn.ceps.eu/wp-content/uploads/2017/01/WD2017-01GT%20FiscalUnion.pdf
- Van Rompuy, Herman, José Manuel Barroso, Jean-Claude Juncker and Mario Draghi. (2012). "Towards a genuine economic and monetary union". Council of the European Union and European Council. https://www.consilium.europa.eu/uedocs/cms_Data/docs/pressdata/en/ ec/134069.pdf
- Vandenbroucke, Frank, and Chris Luigjes. (2016). "Institutional Moral Hazard in the Multi-tiered Regulation of Unemployment and Social Assistance Benefits". CEPS Special Report, 137, Centre for European Policy Studies, European Commission. https://www.ceps.eu/ceps-publications/ institutional-moral-hazard-multi-tiered-regulation-unemployment-and-social-assistance/
- Vroman, Wayne. (2010). The Role of Unemployment Insurance as an Automatic Stabilizer During a Recession. IMPAQ International. https://www.dol.gov/sites/dolgov/files/ETA/publications/ ETAOP2010-10.pdf
- Whittaker, Julie M., and Katelin P. Isaacs. (2014). "Unemployment insurance: Programs and benefits". Congressional Research Service, Washington, DC. https://crsreports.congress.gov/ product/pdf/RL/RL33362
- Zandi, Mark. (2008). "Washington throws the economy a rope". Moody's Analytics. https://www.economy.com/economicview/analysis/102598

A Designing and modelling a EUIS

This section illustrates the main challenges related to the implementation of an EUBS, together with some solutions proposed in the literature. I have focused on studies that simulated the effect of potential EUBS because these studies had to make choices regarding their operational definitions and how some commonly encountered problems were dealt with. Examples of such studies include *Dullien* (2013, 2014), *Italianer and Vanheukelen* (1993), *Beblavý and Maselli* (2014), *Beblavý et al.* (2015), *Jara and Sutherland* (2014) and *Dolls et al.* (2014).

A.1 Designing options

According to the terminology of *Beblavy* et al. (2017), a EUI system for the euro area could be designed in two ways: a so-called "genuine" insurance system, or an "equivalent" system.

- In a "genuine" eurozone unemployment insurance scheme, the insurance is provided directly to individuals in the event of unemployment, temporary transfers made directly by the central authority and there is no national government intervention *Thirion* (2017). The main objective of this scheme is to stabilize household income in the event of a negative shock and it is permanently active (*Beblavý et al.*, 2017). Examples of these proposals are *Dullien* (2013) and *Dolls et al.* (2018).
- An "equivalent scheme" is a reinsurance mechanism for existing national unemployment schemes, in which transfers inter-government are directly related to the unemployment rate but not to output gaps (in contrast with the rainy-day funds, that are linked to output gaps). Using the unemployment rate has several advantages over using the output gap, such as the fact that it is directly observable, more accurate in forecasting, and less subject to adjustment. This mechanism allows transfers between countries unaffected by asymmetric shocks and countries primarily affected by the shock, ensuring risk sharing and enabling stability in the euro area. Transfers are made directly to the government experiencing the negative shock, but not to individuals (Dolls et al., 2018; Thirion, 2017). Typically, this mechanism incorporates a trigger, meaning that a country will receive payment only when the threshold is reached and the mechanism is activated (*Beblavy*) et al., 2017). A low threshold implies that the insurance scheme is triggered frequently and vice versa. Also in this system, the conditions that determine the pay-in and pay-out are similar across the countries. One crucial caveat is how the funds received are used by the government: are these funds earmarked to unemployment benefits or can governments choose how these funds are spent? The stabilisation capacity of the reinsurance scheme depends on this.

Within the set of genuine EUIS, there is a distinction between "basic" and "top-up" schemes. Basic schemes are widely supported, while few studies view top-up schemes as viable (*Delpla*, 2012). The idea behind a basic genuine European unemployment insurance system is to create a scheme that (at least) partly replaces existing national schemes. Benefits are paid out directly to the unemployed, and the same conditions apply to all workers in Europe. National governments can go beyond the

European insurance scheme, for example by extending coverage and generosity. In other words, national governments can top up the basic European insurance scheme.

The "top-up" EUIS is different from the "basic" schemes in that it guarantees a minimum benefit, which includes all dimensions of the insurance. For example, it also refers to the duration of the benefits and the eligibility conditions. When the national scheme is sufficient to meet these requirements, the European scheme does not intervene. In the opposite case, the European scheme covers the difference by topping up the national scheme. In the study of *Delpla* (2012), this idea is translated as follows: workers can choose between the European and the national insurance scheme if the national scheme is more attractive.

These two types of schemes require different degrees of harmonization. Setting up a genuine European unemployment insurance requires a high degree of harmonization, as benefits are transferred directly from supranational authorities to citizens, requiring similar requirements, generosity, and deadlines for all countries. Such a mechanism would be difficult to set up and would require a high level of integration of economic and social policies. An equivalent European unemployment insurance system, or reinsurance system, would be designed as an addition to existing national systems, allowing national governments to fund unemployment benefits without changing national policy, since the scheme transfers to the country the amount needed to pay the benefits according to each national system. This will encourage countries to converge towards more generous national systems without the need for formal harmonization (*Thirion*, 2017). This mechanism, which is close to the US federal unemployment insurance scheme, seems more compatible with the heterogeneous domestic preferences of Member States in terms of social policy.

Focusing on the design of an "equivalent" system, three key options have been discussed in the literature and in the policy debate so far.

- A first option would be a common EMU-UI system that provides a basic level of insurance by partly replacing national unemployment insurance systems. Benefits from the euro area system could be topped up by additional payments from national unemployment insurance systems. Hence, there would be room for diversity across member states so that existing differences with regard to replacement rates and benefit duration could be preserved. The EMU-UI system would be financed by social insurance contributions with a contribution rate that could be uniform across Eurozone member states or country specific and time variant to restrict cross-country transfers. An important feature of such a scheme is that it would provide income insurance to the unemployed (under certain eligibility conditions) irrespective of the size of the unemployment shock in a given member state.
- As an alternative, a common scheme could provide income stabilization only in the event of large (unemployment) shocks. Such contingent unemployment benefits would be triggered if the level and/or change in overall unemployment has reached a pre-determined threshold in a given period. National unemployment insurance systems would still be in place in normal times.

• As a third option, the euro area unemployment insurance scheme could complement national systems by providing additional transfers which would either top up national benefits or kick in if national benefits expire. The payout rules of this scheme could be trigger based as well. Such a system would be comparable to the US unemployment insurance system where regular state benefits can be complemented by two types of benefit extension programs which are at least partly provided by the federal government, the extended benefit program (EB) and emergency benefits.

Both genuine and reinsurance schemes are always composed of a range of features that can take different values. Some of these features are found in national unemployment benefits schemes, while others are generally only present in the European schemes.

- The first thing policy-makers have to consider when designing a EUIS is eligibility. This includes who qualifies for benefits and the conditions that have to be met. In national schemes, eligibility depends on several factors, including the nature of unemployment (voluntary or involuntary), the employment (how long has someone worked, vis-a-vis a reference period) or unemployment record, personal characteristics (e.g. worker or self-employed), etc. Eligibility conditions determine coverage, and hence have an impact on stabilisation. They can be strict or easy to meet.
- A second element is the benefit amount: The amount of benefits that an eligible unemployed worker receives is often calculated on the basis of a reference wage (e.g. gross or net, last or average wage) to which a certain replacement rate and caps are applied. Yet, there are countries in which unemployment benefits are flat-rated.
- A third feature is the duration of the benefits: how long are benefits paid out and does this vary over time. In many national schemes, benefit duration depends on the case (e.g. related to the age of an unemployed individual). Another point is whether there is a waiting period before benefits are paid out.

All these features are used to determine the level and duration of benefits in the genuine schemes and to establish who is eligible. In the reinsurance schemes, these features apply as well: governments receive an amount of transfers that is sufficient to support unemployed individuals who meet the eligibility conditions with a benefit amount that is calculated according to these criteria for a pre-defined duration.

In addition to these features, a EUIS can have experience rating, claw-back and cyclical variability. Experience rating and clawback are two instruments that link the pay-in into the scheme to the use of the scheme. Experience rating works ex ante, while claw-back operates ex post. Experience rating ensures that countries that use the scheme more, or that have a higher probability of doing so, have to pay more into the scheme. Claw-back tackles long-term imbalances vis- a-vis the scheme. If Member States have a negative cumulative balance (i.e. they have received more than they have paid into the fund), the scheme can adjust this situation by increasing the country's pay-in or demanding a supplementary contribution

A.2 Triggering

One of the most critical design options of a EUIS is the trigger mechanism, which is composed of a trigger (or indicator) variable and a threshold. The trigger variable measures the magnitude of fluctuations in employment. It enters the activation rule determining under which circumstances a pay-out from the stabilization fund is triggered. Most proposals rely on indicators such as the unemployment rate or the short-term unemployment rate. A threshold value in the activation rule stipulates how large the size of the shock must be for a pay-out to be triggered. It can either apply to the quarter-on-quarter change in the trigger variable or to the deviation from its long-term moving average.

The first issue is the choice of the trigger variable. To ensure effectiveness, the capacity should be activated based on a transparent, certain and unbiased real-time measure of economic activity with limited ex-post revisions. In theory, the output gap is the most straightforward indicator of cyclical developments. However, several contributions to the literature note that the output gap is not an appropriate measure, regardless of its relationship to the business cycle, as it is unobserved and frequently modified (for example, see Biggs and Mayer, 2010; Ince and Papell, 2013; Strauss et al., 2013; Darvas, 2015).⁸ Furthermore, in a monetary union like the EMU, where the traditional adjustment mechanism (exchange rate) is lost and fiscal policies are constrained, the adjustment burden probably is on unemployment (*Pasimeni*, 2015). On the other hand, an unemployment based trigger appears as a viable pragmatic option for the activation of support. The unemployment rate has several valuable properties: it is well-known, harmonised, available at high frequency with short delays, and subject to limited revisions. It is an excellent indicator of the business cycle, purging some of the short-term noise of GDP. It reacts however with some lag to the business cycle. This may not be such an issue for a stabilisation function focused on large shocks. Moreover, the effects of shocks on public finances also tend to lag the growth cycle and actually to more or less match the unemployment cycle. A more significant potential limitation of the unemployment rate, however, is that its sensitivity to cyclical shocks may differ across Member States, for example because some economies have more developed working-time arrangements in downturns. Another technical consideration is the risk that the assessment of cyclical developments are affected by structural improvements in labour markets, but such effects appear limited empirically.

Table A.1 provides an overview of recent proposals on unemployment-based stabilization schemes. As shown in column 2, most proposals rely on the unemployment rate as indicator variable triggering payouts from the EUI fund (*Arnold et al.*, 2018; *Carnot et al.*, 2017; *Claveres and Stráský*, 2018; *Dullien et al.*, 2017). Other contributions have proposed alternatives such as the short-term unemployment rate (*Beblavý et al.*, 2017), labour market variables capturing both intensive and extensive margin changes, in particular hours worked or the wage bill (*Bénassy-Quéré et al.*, 2018) or GDP (*Gossé et al.*, 2022). Moving next to the activation rule, column 3 reveals that some form of automaticity of payouts is present in all proposals. Payouts are triggered if the indicator variable in the activation rule is above its historical moving average (*Arnold et al.*, 2018; *Beblavý et al.*, 2017;

⁸Still, one has to keep in mind that output gap revisions could also result from changes in the methodology used to estimate them (e.g. methodological improvements) (*Enderlein et al.*, 2013).

Dullien et al., 2017), increasing (*Bénassy-Quéré et al.*, 2018), or only if both conditions are fulled (*Carnot et al.*, 2017; *Claveres and Stráský*, 2018). Threshold values define how large the labour market shock must be for a payout to be triggered.

Author	Trigger variable	Activation rule				
Arnold et al. (2018)	Unemployment rate	Unemployment rate above its 7-years moving average				
Beblavý et al. (2017)	Short-term unemploy- ment rate	Short-term unemployment rate above its 10- years moving average, thresholds: $0.1/1/1$ p.p.				
Bénassy-Quéré et al.	Unemployment rate,	Year-on-year increase in unemployment rate /				
(2018)	employment or wage bill	decline in employment by e.g. 2 p.p.				
Carnot et al. (2017)	Unemployment rate	Double condition: year-on-year increase in un- employment rate and unemployment above its				
		15-years moving average; variants: different thresholds for year-on-year increase				

Table A.1: Trigger mechanisms in the literature

Table A.1: Trigger mechanisms in the literature

Author	Trigger variable	Activation rule
Dullien et al. (2017)	Unemployment rate	1) payment from national compartment: unem-
		ployment rate above its 5-years moving aver-
		age, threshold: 0.2pp. 2) additional payment
		from common department ("stormy day fund"):
		threshold: 2 p.p.
Gossé et al. (2022)	GDP	1) In the event of a recession in a country (two
		consecutive quarters of falling GDP) 2) In the
		event of a 1% fall in national GDP.

Finally, one relevant feature of a trigger mechanism is that it helps to deal with two of the most important intertwined challenges to a EUIS: moral hazard and permanent transfers issues. Moral hazard seems to be inevitable in a monetary union where financial transfers flow from one economic region to another, especially when regions have policy powers (*Grauwe*, 2003). Indeed, a mechanism that supports countries in a downturn may reduce their incentives to carry out reforms. While moral hazard cannot be avoided (*Vandenbroucke and Luigjes*, 2016), there are mechanisms that can be used to mitigate the issue. One mechanism that serves this purpose is a trigger, which has two components: an indicator and a threshold. If a scheme is conditioned by a trigger, it only becomes active in the event of a shock. In other words, when the indicator surpasses the threshold, the EUIS transfers funds to the government. To rule out permanent transfers, (*Beblavý et al.*, 2015) argue that the indicator has to be tied to a rate of change in economic activity (e.g. the unemployment rate, so that every country can benefit) and that the threshold has to be sufficiently high (so that transfer are not permanent by definition).

A.3 Financing and size

A EUIS is also confronted with, at least, another two dimensions: financing and size. When it comes to the financing of the insurance scheme, four sources of finance have been suggested for the EUIS, also on the basis of the experiences of unemployment benefit schemes in federal countries: a payroll tax, a corporate tax, a contribution paid by member countries defined as a percentage of GDP and not linked to a joint ad hoc tax, and debt.

The payroll tax – used in the EUIS proposed by Dullien (2013, 2014) – and the corporate tax (*Pisani-Ferry et al.*, 2013) have the advantage of creating a 'genuine' unemployment benefit scheme, in the sense that they generate a system that works as insurance at a microeconomic level, where the worker or the employer pays a contribution that is directly linked to the assistance the worker will receive in the event of remaining unemployed. These financing schemes would also endow the Commission with a budget that is directly related to a well-defined source of revenue.

In the case of reinsurance schemes, in which transfers flow to the national government when the scheme is triggered (*Italianer and Vanheukelen*, 1993; *Beblavý and Maselli*, 2014; *Dolls et al.*, 2014), the scheme is financed through Member States' contributions. These contributions are expressed as a percentage of the country's GDP, either fixed or variable. This does not mean that payroll and corporate taxes cannot be used as a way to finance an EUBS. In fact, defining the payment as a percentage of a country's GDP is a very general way of defining the country's pay-in, which can coincide with a payroll or corporate tax if the member countries agree that this is the way the national contribution should be collected. In the absence of such an agreement, defining the contribution as a sum relative to GDP leaves member countries free to decide how to collect the necessary resources. In the latter case, the EUBS works as insurance between countries (a so-called 'equivalent system').

Finally, the issue of debt is more controversial, because it would imply the ability of the fund to issue supranational bonds and while debt-issuing has been discussed in many papers, a consensus has yet to be reached on the desirability of this financing channel. Nonetheless, there are important arguments in favour of debt (e.g. *Dullien*, 2013). A first argument is that debt-issuing is essential to deal with large, symmetric shocks, when the scheme likely has insufficient funds. Raising contributions or taxes at this time would put even more pressure on Member States that are already struggling. Such a strategy would weaken stabilisation and is pro-cyclical instead.

In designing an EUIS, it is necessary to think also about the potential size of the scheme in terms of GDP. This is clearly related to the stabilisation effect that one would wish the EUIS to have, which would seem to draw very wide boundaries since different scholars and policy-makers have had very different views in this respect. However, these boundaries are considerably narrowed by the fact that national unemployment benefit schemes are already in place, and it is difficult to

imagine that the EUIS would be much more generous or restrictive than the systems already in place. A simple example of this is given by *Italianer and Vanheukelen* (1993), who estimate the hypothetical annual costs of the EUIS by multiplying the number of unemployed individuals by the average EU wage and then by a replacement rate set at 70%. The result is that the EUIS would cost 0.5% of GDP, which is within the range of costs estimated by later literature.

More recent simulation studies typically set an EUIS with a given duration of unemployment benefits, a replacement rate determining the size of the benefits relative to a reference wage, and a coverage rate (the proportion of unemployed individuals who would receive benefits). These parameters are chosen to be roughly in line with those in place in some EU countries (typically, the median country). Once these parameters are chosen, the size of the EUBS is estimated based on these choices. Most simulation studies set the maximum duration to 12 months (*Dullien*, 2007, 2012, 2013; *Beblavý and Maselli*, 2014; *Dolls et al.*, 2014), the replacement rate at 40% or 50% (*Dullien*, 2007, 2013; *Beblavý and Maselli*, 2014; *Dolls et al.*, 2014; *Jara and Sutherland*, 2014; *Beblavý et al.*, 2015), and the coverage rate at 75% or 80% (*Beblavý and Maselli*, 2014; *Beblavý et al.*, 2015) unless it is computed endogenously according to other rules (*Dullien*, 2013; *Dolls et al.*, 2014).

Dolls et al. (2014) suggest a genuine scheme with no trigger, and the estimated annual cost for the euro area is around \bigcirc 50 billion per year (0.6% of GDP). An upper bound to the EUBS cost is given by ?, who assume that the scheme would cover all the unemployment benefit costs incurred by member countries, which on average for the period 2002-2010 equalled 1.8% of GDP in the euro area. Dullien (2007, 2013) defines different scenarios, with an annual cost varying from 0.75% to 0.85% of euro area GDP (Dullien, 2007) or between 0.3% and 0.6% of EU GDP (Dullien, 2013).

However, some simulation studies incorporate a trigger in their EUIS to limit when the payment is made. Since the claim (the amount sent to a specific country after the conditions are met) still follows the framework of the median country, but the transfers only take place in specific situations, the EUIS is significantly less expensive as a result. For example, under the scenarios in which the trigger conditions are most restrictive, *Beblavý and Maselli* (2014) estimate the average hypothetical cost of their EUIS scheme between 1999 and 2012 to be 0.07% of EU GDP (the average cost is close to 0.3% in their least restrictive scenarios).

Table A.2 summarises the main options in the literature in terms of the size (pay-out) and contributions of the simulated EUIS. All in all, across the literature, estimates of the size of a EUIS appear to vary between 0.3 and 0.85 % of EU GDP. In the US, the cost of regular benefits was estimated to be in the range of \$40.5 billion in 2014, which is equivalent to 0.23% of GDP (*Whittaker and Isaacs*, 2014).

Pay-out rule	Contrubution rule	Borrowing
0.5% of GDP for every 1	0.35% of GDP per year; vari-	Yes
p.p. deviation in the un-	ants: higher/lower contribu-	
employment rate above its 7-	tion rates; experience rating	
years moving average; vari-		
ants: higher/lower transfers		
rates		
	Pay-out rule 0.5% of GDP for every 1 p.p. deviation in the un- employment rate above its 7- years moving average; vari- ants: higher/lower transfers rates	Pay-out ruleContrubution rule0.5% of GDP for every 10.35% of GDP per year; vari- ants: higher/lower contribu- tion rates; experience ratingp.p. deviation in the un- employment rate above its 7- years moving average; vari- ants: higher/lower transfers rates0.35% of GDP per year; vari- ants: higher/lower contribu- tion rates; experience rating

Table A.2: Pay-out and contribution rules in the literature

Author	Pay-out rule	Contrubution rule	Borrowing
Beblavỳ et al. (2017)	Pay-out equals sum of un- employment benefits paid to the short-term unemployed according to the rules of a hy- pothetical genuine EUBS	0.1% of GDP per year until 0.5% of EU GDP is accumu- lated; some variants with ex- perience rating/claw-back	Yes
Bénassy-Quéré et al. (2018)	One-off transfer of a fixed percentage of GDP (0.25%) for each p.p. increase in unemployment/decline in em- ployment beyond the specified threshold	0.1% of GDP per year; experience rating	No
Carnot et al. (2017)	0.5% of GDP per percent in- crease in the unemployment rate, variants: higher pay- outs.	Double condition: year-on- year decrease in unemploy- ment rate and unemployment below its 15-years moving average; variants: different thresholds for year-on-year decrease; 0.5% of GDP de- crease in unemployment; ex- perience rating.	Yes
Dullien et al. (2017)	1) national compartments: 25% of average wages paid per employee 2) common com- partment: transfers becom- ing proportionally bigger the larger the increase in unem- ployment.	0.1% of GDP per year; 80%into national compartment,20% into common compartment; experience rating	Yes
Gossé et al. (2022)	25% of average net wage times the number of unemployed	0.24% of real disposable in- come of households. In addi- tion, when transfers are acti- vated, the contribution rate is set to 0%.	Yes

A.4 Modelling options

Regarding the methodological approaches for simulating a EUI, three approaches have been used to simulate the effects of the introduction of a European unemployment insurance: they concur on the smoothing potential of such a mechanism, despite methodological and calibration differences.

- The first is based on DSGE models.⁹. In a two-country business cycle model with incomplete financial markets and frictional labour markets, Moyen et al. (2019) show that optimal risk-sharing via a supranational unemployment insurance increases the counter-cyclicality of replacement rates compared to the decentralised setting. Simulation results based on a calibration to the euro area's core and periphery suggest significant distributional effects via the un-employment insurance system. Iqnaszak et al. (2020) model a monetary union with a role for a federal unemployment transfer system. As member states control regional labour-market policies, there is a trade-off between stabilisation and moral hazard. Enders and Vespermann (2021) also simulate the effects of a cross-regional fiscal transfer mechanism, where they show how welfare effects depend on different types of shocks. Kaufmann et al. (2023) use a mediumscale DSGE model to quantitatively assess the macroeconomic stabilisation properties of a supranational unemployment insurance scheme. The model is calibrated to the euro area's core and periphery and features a rich fiscal sector, sovereign risk premia and labour market frictions. Adopting both simple policy rules and optimal policies for the centralised insurance scheme, our simulations point to enhanced business cycle synchronisation and inter-regional consumption smoothing. Finally, Abraham et al. (2023) study the welfare effects of both existing and counter-factual European unemployment insurance (UI) policies using a rich multi-country dynamic general equilibrium model with labour market frictions. They find that mechanisms like SURE, which allows national governments to borrow at low interest rates to cover expenditures on unemployment risk, yield sizeable welfare gains.
- A second, more empirical approach, uses microeconomic data to calculate the smoothing effect on household consumption of a European unemployment insurance system, as well as the redistributive effects of such a system between individuals and between countries. By simulating a replacement rate of about 70% partly insured at the European level, *Dolls et al.* (2018) manage to smooth income fluctuations at the European level by 10% on average between 2000 and 2013, and obtain very variable gains across countries. Such an approach allows for a fine-grained analysis of individual gains, but masks general equilibrium effects.
- The third approach uses macroeconometric simulation models to assess the effect of a European unemployment insurance, which allows for a quantitative analysis of general equilibrium effects and feedbacks between countries. *Fichtner and Haan* (2014), for example, study an insurance system close to a true system using the NiGEM model in which they assume that a part of the unemployment expenditure is financed at the European level, thus increasing national public expenditure by the same amount. They show that harmonising and taking over part of the national unemployment insurance systems at the euro area level would have significantly reduced the drop in GDP during the crisis in some countries.

 $^{^{9}}$ Dynamic Stochastic General Equilibrium models are used to assess the macroeconomic impact of monetary or fiscal policy

Additional figures and tables Β



Figure B.1: Pay-outs, contributions and cumulative balance by country (Billions)





Source: Author's own calculations based on NiGEM model



Figure B.2: Counterfactual gain/losses in consumption, GDP and unemployment (%)

(a) Trigger 1



Source: Author's own calculations based on NiGEM model

BANCO DE ESPAÑA PUBLICATIONS

WORKING PAPERS

- 2241 JULIO GÁLVEZ and GONZALO PAZ-PARDO: Richer earnings dynamics, consumption and portfolio choice over the life cycle.
- 2242 MARINA DIAKONOVA, CORINNA GHIRELLI, LUIS MOLINA and JAVIER J. PÉREZ: The economic impact of conflict-related and policy uncertainty shocks: the case of Russia.
- 2243 CARMEN BROTO, LUIS FERNÁNDEZ LAFUERZA and MARIYA MELNYCHUK: Do buffer requirements for European systemically important banks make them less systemic?
- 2244 GERGELY GANICS and MARÍA RODRÍGUEZ-MORENO: A house price-at-risk model to monitor the downside risk for the Spanish housing market.
- 2245 JOSÉ E. GUTIÉRREZ and LUIS FERNÁNDEZ LAFUERZA: Credit line runs and bank risk management: evidence from the disclosure of stress test results.
- 2301 MARÍA BRU MUÑOZ: The forgotten lender: the role of multilateral lenders in sovereign debt and default.
- 2302 SILVIA ALBRIZIO, BEATRIZ GONZÁLEZ and DMITRY KHAMETSHIN: A tale of two margins: monetary policy and capital misallocation.
- 2303 JUAN EQUIZA, RICARDO GIMENO, ANTONIO MORENO and CARLOS THOMAS: Evaluating central bank asset purchases in a term structure model with a forward-looking supply factor.
- 2304 PABLO BURRIEL, IVÁN KATARYNIUK, CARLOS MORENO PÉREZ and FRANCESCA VIANI: New supply bottlenecks index based on newspaper data.
- 2305 ALEJANDRO FERNÁNDEZ-CEREZO, ENRIQUE MORAL-BENITO and JAVIER QUINTANA: A production network model for the Spanish economy with an application to the impact of NGEU funds.
- 2306 MONICA MARTINEZ-BRAVO and CARLOS SANZ: Trust and accountability in times of pandemic.
- 2307 NATALIA FABRA, EDUARDO GUTIÉRREZ, AITOR LACUESTA and ROBERTO RAMOS: Do renewable energies create local jobs?
- 2308 ISABEL ARGIMÓN and IRENE ROIBÁS: Debt overhang, credit demand and financial conditions.
- 2309 JOSÉ-ELÍAS GALLEGOS: Inflation persistence, noisy information and the Phillips curve.
- 2310 ANDRÉS ALONSO-ROBISCO, JOSÉ MANUEL CARBÓ and JOSÉ MANUEL MARQUÉS: Machine Learning methods in climate finance: a systematic review.
- 2311 ALESSANDRO PERI, OMAR RACHEDI and IACOPO VAROTTO: The public investment multiplier in a production network.
- 2312 JUAN S. MORA-SANGUINETTI, JAVIER QUINTANA, ISABEL SOLER and ROK SPRUK: Sector-level economic effects of regulatory complexity: evidence from Spain.
- 2313 CORINNA GHIRELLI, ENKELEJDA HAVARI, ELENA MERONI and STEFANO VERZILLO: The long-term causal effects of winning an ERC grant.
- 2314 ALFREDO GARCÍA-HIERNAUX, MARÍA T. GONZÁLEZ-PÉREZ and DAVID E. GUERRERO: How to measure inflation volatility. A note.
- 2315 NICOLÁS ABBATE, INÉS BERNIELL, JOAQUÍN COLEFF, LUIS LAGUINGE, MARGARITA MACHELETT, MARIANA MARCHIONNI, JULIÁN PEDRAZZI and MARÍA FLORENCIA PINTO: Discrimination against gay and transgender people in Latin America: a correspondence study in the rental housing market.
- 2316 SALOMÓN GARCÍA: The amplification effects of adverse selection in mortgage credit suply.
- 2317 METTE EJRNÆS, ESTEBAN GARCÍA-MIRALLES, METTE GØRTZ and PETTER LUNDBORG: When death was postponed: the effect of HIV medication on work, savings and marriage.
- 2318 GABRIEL JIMÉNEZ, LUC LAEVEN, DAVID MARTÍNEZ-MIERA and JOSÉ-LUIS PEYDRÓ: Public guarantees and private banks' incentives: evidence from the COVID-19 crisis.
- 2319 HERVÉ LE BIHAN, DANILO LEIVA-LEÓN and MATÍAS PACCE: Underlying inflation and asymmetric risks.
- 2320 JUAN S. MORA-SANGUINETTI, LAURA HOSPIDO and ANDRÉS ATIENZA-MAESO: The numbers of equality regulation. Quantifying regulatory activity on non-discrimination and its relationship with gender gaps in the labour market.
- 2321 ANDRES ALONSO-ROBISCO and JOSÉ MANUEL CARBÓ: Analysis of CBDC Narrative of Central Banks using Large Language Models.
- 2322 STEFANIA ALBANESI, ANTÓNIO DIAS DA SILVA, JUAN F. JIMENO, ANA LAMO and ALENA WABITSCH: New technologies and jobs in Europe.
- 2323 JOSÉ E. GUTIÉRREZ: Optimal regulation of credit lines.
- 2324 MERCEDES DE LUIS, EMILIO RODRÍGUEZ and DIEGO TORRES: Machine learning applied to active fixed-income portfolio management: a Lasso logit approach.

- 2325 SELVA BAHAR BAZIKI, MARÍA J. NIETO and RIMA TURK-ARISS: Sovereign portfolio composition and bank risk: the case of European banks.
- 2326 ANGEL-IVAN MORENO and TERESA CAMINERO: Assessing the data challenges of climate-related disclosures in european banks. A text mining study.
- 2327 JULIO GÁLVEZ: Household portfolio choices under (non-)linear income risk: an empirical framework.
- 2328 NATASCHA HINTERLANG: Effects of Carbon Pricing in Germany and Spain: An Assessment with EMuSe.
- 2329 RODOLFO CAMPOS, SAMUEL PIENKNAGURA and JACOPO TIMINI: How far has globalization gone? A tale of two regions.
- 2330 NICOLÁS FORTEZA and SANDRA GARCÍA-URIBE: A Score Function to Prioritize Editing in Household Survey Data: A Machine Learning Approach.
- 2331 PATRICK MACNAMARA, MYROSLAV PIDKUYKO and RAFFAELE ROSSI: Taxing consumption in unequal economies.
- 2332 ESTHER CÁCERES and MATÍAS LAMAS: Dividend Restrictions and Search for Income.
- 2333 MARGARITA MACHELETT: Gender price gaps and competition: Evidence from a correspondence study.
- 2334 ANTON NAKOV and CARLOS THOMAS: Climate-conscious monetary policy.
- 2335 RICARDO BARAHONA, STEFANO CASSELLA and KRISTY A. E. JANSEN: Do teams alleviate or exacerbate the extrapolation bias in the stock market?
- 2336 JUAN S. MORA-SANGUINETTI and ANDRÉS ATIENZA-MAESO: "Green regulation": A quantification of regulations related to renewable energy, sustainable transport, pollution and energy efficiency between 2000 and 2022.
- 2401 LAURA HOSPIDO, NAGORE IRIBERRI and MARGARITA MACHELETT: Gender gaps in financial literacy: a multi-arm RCT to break the response bias in surveys.
- 2402 RUBÉN DOMÍNGUEZ-DÍAZ, SAMUEL HURTADO and CAROLINA MENÉNDEZ: The medium-term effects of investment stimulus.
- 2403 CLODOMIRO FERREIRA, JOSÉ MIGUEL LEIVA, GALO NUÑO, ÁLVARO ORTIZ, TOMASA RODRIGO and SIRENIA VAZQUEZ: The heterogeneous impact of inflation on households' balance sheets.
- 2404 JORGE ABAD, GALO NUÑO and CARLOS THOMAS: CBDC and the operational framework of monetary policy.
- 2405 STÉPHANE BONHOMME and ANGELA DENIS: Estimating individual responses when tomorrow matters.
- 2406 LAURA ÁLVAREZ-ROMÁN, SERGIO MAYORDOMO, CARLES VERGARA-ALERT and XAVIER VIVES: Climate risk, soft information and credit supply.
- 2407 JESÚSFERNÁNDEZ-VILLA VERDE, JOËL MARBET, GALO NUÑO and OMAR RACHEDI: Inequality and the zero lower bound.
- 2408 PABLO BURRIEL, MAR DELGADO-TÉLLEZ, CAMILA FIGUEROA, IVÁN KATARYNIUK and JAVIER J. PÉREZ: Estimating the contribution of macroeconomic factors to sovereign bond spreads in the euro area.
- 2409 LUIS E. ROJAS and DOMINIK THALER: The bright side of the doom loop: banks' sovereign exposure and default incentives.
- 2410 SALOMÓN GARCÍA-VILLEGAS and ENRIC MARTORELL: Climate transition risk and the role of bank capital requirements.
- 2411 MIKEL BEDAYO and JORGE E. GALÁN: The impact of the Countercyclical Capital Buffer on credit: Evidence from its accumulation and release before and during COVID-19.
- 2412 EFFROSYNI ADAMOPOULOU, LUIS DÍEZ-CATALÁN and ERNESTO VILLANUEVA: Staggered contracts and unemployment during recessions.
- 2413 LUIS FÉRNANDEZ LAFUERZA and JORGE E. GALÁN: Should macroprudential policy target corporate lending? Evidence from credit standards and defaults.
- 2414 STÉPHANE BONHOMME and ANGELA DENIS: Estimating heterogeneous effects: applications to labor economics.
- 2415 LUIS GUIROLA, LAURA HOSPIDO and ANDREA WEBER: Family and career: An analysis across Europe and North America.
- 2416 GERALD P. DWYER, BILJANA GILEVSKA, MARÍA J. NIETO and MARGARITA SAMARTÍN: The effects of the ECB's unconventional monetary policies from 2011 to 2018 on banking assets.
- 2417 NICOLÁS FORTEZA, ELVIRA PRADES and MARC ROCA: Analysing the VAT cut pass-through in Spain using webscraped supermarket data and machine learning.
- 2418 JOSÉ-ELÍAS GALLEGOS: Hank beyond fire: Amplification, forward guidance, and belief shocks.
- 2419 DANIEL ALONSO: Stabilisation properties of a SURE-like European unemployment insurance.